

Partner's income shock and female labor supply. Evidence from the repeal of Argentina's convertibility law

Laurine Martinoty

► To cite this version:

Laurine Martinoty. Partner's income shock and female labor supply. Evidence from the repeal of Argentina's convertibility law. World Development, 2022, 159, pp.106039. 10.1016/j.worlddev.2022.106039 . hal-03767916

HAL Id: hal-03767916 https://hal.science/hal-03767916v1

Submitted on 7 Sep 2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

Partner's Income Shock and Female Labor Supply. Evidence from the Repeal of Argentina's Convertibility Law

Laurine Martinoty*

Abstract

Female employment is an important vector of economic development. Using data on couples in urban Argentina from 1996 to 2007, I show that in the short and medium term necessity shapes female participation and employment at the extensive and intensive margins. More specifically, I study how women's labor supply reacts to negative income shocks affecting their partner. In order to assess the causal impact, I exploit the unexpected evolution of the economic environment triggered by the repeal of the convertibility law in January 2002 to instrument men's job loss. I find that women's probability of participating and finding a job is multiplied by 2 upon their partner's displacement. Turning to the dynamics of their labor supply, contrary to expectations, however, women do not symmetrically withdraw from the labor market once their partner finds a job. Evidence on repeated cross-sections confirms that the labor supply response persists long after the economic recovery. My findings are among the first attempts to evaluate the participation effects of temporary shocks in the medium term.

Keywords: female labor force, female employment, intra-household allocation

JEL: D10, J22, O12, O54

Highlights:

- Female employment is an important vector of economic development.
- This paper uses Argentine panel data (1996-2007) and studies how women's labor outcomes react to negative income shocks affecting their partner.
- The sudden repeal of convertibility in Jan. 2002 caused an exogenous change in male employment serving as an instrument for their actual job loss.
- The female labor supply response is very high: when unemployment hits her partner, a woman is twice as likely to enter the labor market and work.

^{. **} University of Paris 1 Pantheon-Sorbonne - CES (8174); Correspondance: laurine.martinoty@univparis1.fr. Maison des sciences économiques, Office 303, 106-112 bd de l'Hôpital, 75013 Paris, France.

• This effect persists over several years, suggesting that temporary labor supply adjustments can develop an attachment to the labor market.

Declarations of interest: none.

1 Introduction

One of the most striking differences across developing regions relates to the participation of females in the labor force. Between 1990 and 2019 in Latin America and the Caribbean, the participation of women aged 15-64 increased sharply from 44.2 to 57.9%. Over the same period, it stagnated at a high level in Sub-Saharan Africa (63.6 to 62.8%), and at a much lower level in the Middle-East and North Africa (18.4 to 21.2%), while it decreased by 5-6 percentage points in Asia (from 71.8 to 66.2% in East Asia and the Pacific, and from 30.5 to 25.2% in South Asia).¹ As revealed by Duflo (2012), a strong body of evidence suggests that female employment matters for women's well-being and empowerment, and ultimately for many layers of development. In consequence, a large and growing literature is analyzing female participation in the labor force of the developing world, with the aim of capturing its diversity, and designing suitable policies.

Recently, the literature on female participation has put particular emphasis on the interplay between the structural transformation of an economy, on the one hand, and, on the other, the enduring gender norms against women's working, deeply rooted in preindustrial economic conditions (Heath and Jayachandran, 2017; Giuliano, 2017; Jayachandran, 2020). This growing strand of literature essentially focuses on the determinants of the *medium term* dynamics of women's labor force participation. Therefore, it indirectly overlooks the effect of *temporary* negative income shocks on women's choosing to work, notably the possibility that women may also enter the labor force by necessity. In this paper, I contribute to the literature showing that necessity can be an important driver for female participation in developing countries, even when they are at an advanced stage of development. Interestingly, I show that such short term shock coping responses also have persistent effects on female participation in the medium term.

First, I rely on labor force panel surveys representing urban Argentina over the period 1996-2007, and I explore the within-household correlation patterns between partners' labor outcomes, controlling for local labor market time-specific effects. Women whose partner becomes unemployed are 12% more likely than other women to enter the labor force, 8.5% more likely to be employed, and 6.4% more likely to work more than 20 hours a week. At the intensive margin, working women are 23% more likely to declare that they would like to work more, though they do not manage to increase their working time.

In the core of the paper, I set up an instrumental variable strategy to pin down the causal impact of a partner's job loss on female participation. In March 1991, the highly popular convertibility law began to peg the Argentine peso to the dollar, and successfully contained hyperinflation, stimulated growth, and also restored trust in financial institu-

tions. When the Argentinean economy slid from a recession into a depression in 2001, jeopardizing the sustainability of this exchange regime, 'the authorities were unwilling even to consider the possibility of an exit: neither the government nor the public were prepared to take such a drastic course until it was forced upon them by events' (Daseking et al., 2005a). Still, the unsustainable convertibility law finally had to be repealed in January 2002, after two months of unprecedented economic, social and political turmoil. I measure the unexpected decline in male employment following the sudden collapse of the convertibility regime for each of the 88 industry-occupation pairs in the data, and use it as an exogenous source of variation for the actual job loss experienced by male partners between 2001 and 2002. In support of this empirical strategy, I document how a series of events built up to bring convertibility to an unexpected end, and I show that the differential effect of this shock on male unemployment across the different industry-occupation pairs could not be anticipated by looking at the employment trends before the collapse. I also show that because industries and occupations tend to be segregated by gender, the shocks experienced by men and by women do not correlate within couples. Using data on Argentinean couples between October 2000 and October 2002, I confirm that my measure for the shock intensity on male employment is positively and closely correlated with the probability that male partners will actually lose their job. As a main result, I find that an unanticipated shock on male employment multiplies by 2 the probability that their partner will join the workforce for at least one hour per week. Unraveling this result, I show that half of this effect is driven by the rapid expansion of the unemployment assistance program Jefes y jefas which relaxed the constraints on labor demand.

Of course, displacement is not the only source of the negative income shocks incurred by households. In the face of inflation, real wages were also affected in different ways across industries and occupations, because the differential rise in unemployment across industries and occupations translated into a differential power to bargain over the nominal wages for workers. I empirically confirm the link between the magnitude of the employment shock and the decline in real monthly wages. I find evidence that a negative income shock on male workers also generated a positive labor supply response from their partner, concentrated on the bottom part of the pre-crisis household income distribution.

Finally, within the limits of my data, I question further the dynamics of female participation in the presence of negative shocks on male employment. An interesting and new finding is that female participation reacts to job *losses*, not job *entries*: women do not withdraw from the labor market once their partner finds a job. Furthermore, I find that the participation response is for most part contemporaneous with the shock, but may also be partly delayed to the following period. Corroborating these two results on repeated cross-sections, I estimate that the female participation response to the 2002 economic shock persisted for at least three years – and possibly even to the end of my period of observation, albeit on a smaller scale. Taken together, these results indicate that a shock coping response can have persistent effects on female participation, possibly through the channel of labor market attachment.

This paper makes a twofold contribution. First, causal estimates of the response of female workers to negative income shocks affecting their partners in developing countries are lacking. A number of empirical studies investigate the timing of partners' transitions from employment to unemployment in the US (Stephens, 2002; Juhn and Potter, 2007), as well as in Europe (Bredtmann et al., 2018; Halla et al., 2020). Overall, they find women's employment response to male unemployment to be moderate at most. In the US, according to Stephens (2002), women would be discouraged from working, since shocks are likely to be correlated within households. Following Cullen and Gruber (2000), the generosity of social programs would crowd out intra-family insurance. Andersen et al. (2021) compare alternative coping mechanisms in response to male job loss in Denmark, and find that adjustment occurs mainly through reductions in savings and spending. Given the fundamental institutional, social, and economic differences between developed and developing countries, for instance, in terms of access to a social safety net, or to the credit market, the external validity of these results is essentially confined to the developed world. Through being set in the context of an emerging economy, this article contributes to the literature by offering an original empirical investigation of women's labor supply response to negative shocks to their partners' income.

So far, the literature on female labor market responses to negative income shocks in developing countries has essentially been descriptive. Cross-country regression estimates suggest that in upper-middle-income economies like Argentina, the increase in women's participation in response to economic and financial crises is moderate at most. Using a sample of 17 middle income economies, Cho and Newhouse (2013) find little evidence that women's employment responded to the shock to men's employment during the Great Recession. Signorelli et al. (2012) document the impact of financial crises on female participation using a large panel of countries between 1980 and 2005. They find that financial crises *negatively* affect women's participation in upper-middle-income economies, but their data do not allow to study the heterogeneity of women's response according to their partner's exposure to crises. Other findings focus on particular developing countries. Using the Indonesia Family Life panel, Frankenberg et al. (2003) document the immediate effects of the Asian crisis on the well-being of Indonesian families. They provide detailed descriptive evidence that the household labor supply increased at both the extensive and

intensive margins during the Asian financial crisis, but they do not show that this response correlates with their local measure of the magnitude of the economic shock. Studying the Mexican peso crisis with a pseudo-panel, McKenzie (2003) finds that the coping strategy of adding more household members to the labor force, or increasing the labor hours of members already working, is not widely used. However, due to the cross-sectional nature of the data, he cannot rule out that the labor response may in fact vary with the intensity of the shock.

Within this literature, a few papers focus more specifically on Argentina. McKenzie (2004) investigates whether households are able to mitigate the adverse income effect brought by the 2002 financial crisis. Among his findings, he shows that labor supply alone does not allow to compensate for the loss in income. However, he notes that females entering the workforce contribute as much as males entering it, or more. Closest to my paper, Cerrutti (2000) establishes a direct connection between the growth in female labor force participation in Buenos Aires between 1991 and 1994, and male job instability within households. Using a first difference estimation, she documents that women living with partners whose labor force status changed are twice as likely to enter the labor force as those who live in households where the partner is continuously either employed, or out of the labor force. However, she remains inconclusive about whether her results are driven by job entries, job exits, or job instability in general. More to the point, the strength of her results is greatly hampered by the lack of an instrument to account for the endogeneity of partners' labor supply decisions. Using data on the Argentinean labor market over a longer time-frame (1996-2007) and a wider geographical area (the whole of urban Argentina), I am able to provide such an instrument, which allows for a causal measure of the female labor market's response to a partner's unemployment or income shock, as well as for a broad discussion on the mechanisms in play.

To the best of my knowledge, Ayhan (2018) proposes the only other attempt to capture the causal effect of male unemployment on female participation in the context of a developing country. Using EU-SILC panel data from Turkey in the period 2007-2010, she instruments the probability of men's losing their job with variations in the output of male-dominated sectors during the 2008 crisis. She finds that the participation of Turkish married women increased by 24 percentage points in response to their partner's unexpected job loss. However, she does not study other margins, such as employment, or the hours worked, since she concentrates on the subsample of couples where the women were inactive before the Great Recession. By contrast, my paper focuses on all the women in couples, and provides evidence on a wider range of female labor outcomes, in a very different cultural area, where female participation is already high. As in her paper, my empirical strategy also makes use of the gender segregation across industries; however, while her source of exogenous variation for male unemployment stems from the variation in output from seven broad economic sectors, my instrument captures labor demand shocks at a fine industry-by-occupation level (22×4 cells), and my results are robust to the inclusion of province-by-period fixed effects, as well as to clustering the standard errors at the industry-by-occupation level.

With this paper, I also contribute to the broader literature on female labor supply in developing countries. First, in a comparative study of eight developing countries since 2000, Klasen et al. (2021) observe that in two upper-middle-income countries (South Africa and Brazil), households' economic conditions, and notably household income, do not correlate with married women's participation. They conclude that the labor supply choices of urban women in the richest developing countries are not very sensitive to income effects, following a pattern also observed for women in the US (Blau and Kahn, 2007). My main results mitigate this conclusion: I find that women respond strongly to unexpected shocks affecting their partner's labor income. My results are consistent with the effects of other types of shocks on female employment, such as political conflicts. For instance, studying the civil conflict in Nepal from 1996 to 2006, Menon and Van der Meulen Rodgers (2015) show that women who lived in high-conflict areas engaged in more work during the civil war compared to women living in low-conflict areas, in response to their husbands' loss of income caused by war-related disruptions. In addition, beyond this core result, I collect a range of evidence showing that female participation in response to temporary income shocks may in fact be long-lasting. Indeed, women enter the labor force when their partner loses his job, but do not become symmetrically inactive once their partner returns to work. I also find that the female participation response to the 2002 shock experienced by their partner persists at least until 2005, and possibly even to the end of 2007. I conjecture that, to a certain extent, even temporary labor supply adjustments can develop an attachment to the labor market. This conclusion is original, since it implies that temporary shocks need not necessarily be contrasted with underlying trends when one seeks to understand the evolution of the female labor supply. My paper also adds to the existing evidence on female labor force participation in Latin America. Using macro ILO estimates for female participation in various developing regions and countries over the period 1990-2015, Klasen (2019) notes that the decline in fertility in this developing region and the bridging of the gender gap in education correlate particularly strongly with female participation. In terms of female participation, Latin American countries are generally considered good performers when compared to other developing regions, which may explain why they are relatively under-studied (Gasparini and Marchionni, 2015). My paper contributes to filling this gap.

The paper is organized as follows. Section 2 provides background information regarding the convertibility law and the socioeconomic context. Section 3 describes the data. Section 4 discusses the identification strategy. Section 5 presents my findings. Section 6 discusses the robustness of my findings, and investigates the underlying mechanisms. Section 7 questions the dynamics of female participation in the presence of temporary shocks. Section 8 concludes. The Appendix contains further data details and robustness checks.

2 Background Information

In this section, I provide background information regarding the convertibility law, pointing to the fact that the collapse of the convertibility regime was mostly unanticipated by households before autumn 2001. I also provide general information about employment protection in Argentina around 2002, highlighting the weaknesses of the unemployment insurance system, and the importance of an unemployment assistance program introduced in January 2002.

The convertibility law – The convertibility law was adopted in March 1991 under the presidency of Carlos Menem, with the objective of containing hyperinflation. The law provided that the newly created Argentinean peso would be pegged to the dollar, at a oneto-one exchange rate. In parallel, a series of structural reforms were undertaken in order to ensure the credibility and sustainability of this new exchange regime. In the spirit of the Washington Consensus promoted by the IMF, these reforms included massive waves of privatization for publicly owned companies, as well as measures for the liberalization of trade, and of the labor market. These liberal reforms formed the backbone of the Argentinian economy during the presidency of Carlos Menem, or *menemato* (1989-1999). They contributed to restoring people's trust in the domestic currency: prices had been growing by 1300% between January and December 1990, but in April 1991, inflation shrank to 5.5% monthly, and by April 1992 the annual inflation rate was reduced to 25%. The Argentine economy grew at an annual rate of 5-10% over the period, and weathered the contagion of the Tequila crisis with success in 1995. Although the reforms came at the cost of growing unemployment (Cerrutti, 2000), the convertibility remained extremely popular among the middle class.

A moderate recession - In 1999, Argentina entered a moderate recession because of a combination of international and domestic factors. The Russian financial crisis, the US monetary policy, and the currency devaluation in Brazil led to a first phase of destabilization (Fanelli, 2002). Furthermore, 1999 was presidential election year in Argentina. A spending race between the incumbent president and his opponents in the Peronist party jeopardized fiscal austerity. The opposition (a center-left coalition) won the elections, but the newly elected president, Fernando de la Rúa, lacked the necessary political support (Corrales, 2002). This combination of economic and political factors built up vulnerabilities in the convertibility regime. In spite of the fiscal deterioration in 1999 and 2000, however, Argentina was repeatedly backed by IMF-supported programs.

So when did people lose their trust in convertibility? Although Argentinians had been used to booms and busts since WWII, they had serious reasons to believe that this time was different. First, the convertibility was enshrined in law. Then, under the convertibility, the country had experienced an unusually prolonged period of sustained growth with no inflation. Finally, until shortly before the crisis, the country had been publicly praised by the IMF and the US Treasury for its achievements in stabilization, economic growth and liberal reforms. In spite of the recession, Argentina, like any other emergent economy, was predicted to recover rapidly (Outlook, 1999).

Doubts and commitment – More serious doubts about the stability of the currency board did not surface until 2001 (Daseking et al., 2005b). Early in 2001, the first signs of pessimism over the sustainability of the regime came from the capital markets, as the spread on Argentine over US bonds suddenly rose (IMF, 2003). During the first half of 2001, the Argentine authorities reiterated their commitment to severe fiscal austerity and the convertibility regime with several unpopular measures, demonstrating that they were willing to save convertibility at all costs, including electoral ones. In these efforts, they were publicly backed by the IMF with massive loans. In particular, in July and August, the outflow of bank deposits increased as a consequence of a mixture of fears among savers and investors. The joint response of the government with the 'deficit zero' law (July 29th) and the IMF with a loan (August 22nd) temporarily restored trust: the main index of the Buenos Aires stock exchange rose, the investment risk rate in Argentina fell (Relea, 2001), and in September, bank cash deposits were increasing again (Otaloa, 2001). Mid-term elections took place in October 2001. The government issued a statement saying that it would maintain its policy of economic austerity to maintain convertibility, regardless of the outcome of the mid-term elections (Illiano, 2001). When asked about economic policy, the main political opponent of the government coalition, Eduardo Duhalde, declared '... it is the model that is perverse' (Naranjo, 2001), but ruled out dollarization and did not mention devaluation at all. On this occasion, voters harshly punished the political establishment, by abstaining or submitting blank or defaced ballot

papers. The main sources of discontent were not convertibility per se, but corruption and discontent with the general course of public affairs, notably the management of the recession and the budgetary cuts (Corrales, 2002). This debilitated even more the already fragile legitimacy of the incumbent government.

The end of convertibility – In November 2001, the situation rapidly escalated. A bank run started, and the government's response came in the first days of December, in the form of the highly unpopular *corralito* measure of freezing bank accounts. Shortly afterwards, the IMF announced the end of its support, due to Argentina's inability to meet the loan conditions on the zero deficit. This triggered the social and political collapse of December 2001, as well as the end of convertibility. Protests in the form of *cacerolazos* were so massive and violent that President Fernando de la Rúa had to resign; he escaped from the presidential palace by helicopter on December 20, 2001. Note that even in the midst of these events, the IMF predicted a small 1.1% fall in output and deflation for 2002 (Outlook, 2001), suggesting that experts still contemplated the possibility that convertibility would survive the economic, social and institutional collapse of December 2001. But on January 6th, 2002, the newly appointed president Eduardo Duhalde repealed the convertibility law that had been in place for ten years. The inflation rate rose at once from 0 to 10% per month, the recession reached 10% of GDP in 2002, and the unemployment rate peaked at 22%.

Employment protection policies – When the economic crisis of 2002 broke out, Argentina had two main employment protection policies at the national level beyond the severance payment introduced in 1974: a passive unemployment insurance (UI), and an unemployment assistance program (UA). Introduced in 1991, the UI takes the form of a monthly income support restricted to formal sector workers who contributed to the Fondo Nacional de Empleo at least six months during the past three years, and excludes certain sectors, such as agriculture, domestic services, public administration, or private education. Because access to UI is very limited, the share of unemployed receiving payments through the unemployment insurance remained between 5 and 7.5% between 1999 and 2005 (Iturriza et al., 2011). In addition, the benefit duration is short (from 2 to 12 months, according to the length of the contribution), the initial replacement rate is low (on average 40% between 1998 and 2005), and decreases with the duration of unemployment (Bertranou and Paz, 2007). By contrast, UA programs offer a greater coverage of the unemployed population. The first large scale UA program was introduced in 1996. Trabajar provided a cash assistance of up to 200AR\$ a month in exchange for participation into projects of social interest for 33 hours a week (Bertranou and Paz,

2007). This emblematic program ended in 1999, but programs of this type have since multiplied, including *Crear Trabajo* (2000), or *Emergencia Laboral* (2000). Introduced in January 2002 under loan and technical assistance from the World Bank, the workfare program *Jefes y jefas* is certainly the best illustration. Eligible for entry to the program were unemployed household heads with at least one child under 18. Though the program was universal, 20 weekly working hours were required as a counterpart for the 150AR\$ (representing around half of the mean household income per capita in 2002), in order to target the poorest households, whose members have a lower reservation wage. It was extended after April 2002, and reached more than 2 million beneficiaries within a few months.

3 Data

The data stem from the Encuesta Permanente de Hogares (EPH) collected by the Instituto Nacional de Estadisticas y Censos (INDEC). Since 1996, the EPH has been a representative survey of households living in the main urban areas of Argentina. The survey provides detailed information on various labor outcomes and the usual socioeconomic characteristics at the household and individual level. Each household is observed at most four times, then rotated out. Before 2003, the survey was conducted in May and October of each year, and households were followed over 18 months. In 2003, the EPH Continua replaced the EPH Puntual. The survey became quarterly, with households observed for two years in succession, for the same two consecutive quarters of each year.

Main sample – I use all the *EPH* waves surrounding the 2002 crisis, from the first nation-wide survey to the Great Recession (1996-2007).² My main population of interest comprises women and men living together as household head and partner.³ Bearing in mind legal age restrictions regarding marriage and retirement, I consider couples where women and men are aged 18-60 and 18-65 respectively (81.4% of observations). Because I am concerned with female labor market responses to shocks on male labor outcomes, I focus on active men (93.7%) with duly reported labor outcomes (96%). Since my empirical approach exploits within-household variations in time, I drop singletons (5,585 observations). My final sample consists of 243,240 observations on 81,956 unique couples, observed at least twice over the period 1996-2007.

Descriptive statistics – Table 1 documents the labor outcomes of women and men of my main sample over three periods.⁴

[Table 1 about here]

Female participation and employment increase sharply over time, from 47% to 58%, and from 41% to 52% respectively. In line with expectations, unemployment for men peaks between 2000 and 2002. At the intensive margin, employed individuals report the number of hours that they worked per week and state whether they would like to work more. This latter question is my measure for underemployment, equal to 1 if they answer positively to this question, 0 otherwise. As shown in Table 1, at the intensive margin, working women do not increase their hours of work, and underemployment decreases by one third between 2000-2002 and 2003-2007. For both genders, but mostly for men, monthly real labor income⁵ decreases over time, and the simultaneous decrease in hours worked does not fully account for this decline. As a consequence, the hourly wage of women and men tend to converge. Tables A1 and A2 in the Appendix also display standard statistical information on individuals and households, and document their other income sources beyond the labor income of both partners. Table A1 shows that, on average, the women are younger than the men, and educated longer. The level of education increases in time for both sexes, while the number of children per couple decreases. As shown in Table A2, the primary additional source of income for one in every five couples is labor income from other household members. Pensions, unemployment benefits, or capital income also represent non negligible amounts, but concern very few households.

4 Identification

This section presents and discusses the baseline specification, and proposes an instrumental variable model to overcome its limitations.

Baseline specification – My objective is to measure a woman's response in terms of labor supply to a shock in her partner's employment. The empirical specification is given by:

$$Y_{h(r),t}^{f} = \gamma Unemployment_{h,t}^{m} + X_{h,t}\boldsymbol{\beta} + c_h + \mu_{r,t} + \epsilon_{h(r),t}$$
(1)

where $Y_{h(r),t}^{f}$ is the labor market participation of a female partner f, in household h living in region r at period t, and $Unemployment_{h,t}^{m}$ is a dummy equal to one if her male partner m is unemployed.⁶ The specification controls for household fixed effects c_h . This ensures that unobserved differences across households, such as the cognitive skills or soft skills of both spouses, which can affect employment and participation, do not bias the estimates. I therefore identify the parameter of interest γ on variations in labor market outcomes within rather than across couples.⁷ The specification also includes a set of period-byprovince fixed effects $\mu_{r,t}$, which accounts non-parametrically for any common variations in the female participation in the labor market within provinces over time. I therefore fully absorb any shock to female participation at the province level, such as local wage shocks, variations in the local cost of living, or local changes in gender norms.⁸ The specification also controls for a series of time-varying covariates at the household level. A first set of controls measures the evolution in the household composition: the number of infants (0-2), pre-school children (3-5), children (6-15), enrolled children over 15, adults aged 16-64 and adults over 65, as listed in the bottom part of Table A1. I also include as controls the labor income of the household (excluding partners'), as well as its non-labor income, as listed at the bottom of Table A2; i.e. pensions, capital income, unemployment benefits, remittances, and other non labor income. Note that obvious reasons exist for believing that these covariates may be endogenous to female participation. For instance, the presence of a grand-parent can foster the participation of the main female caregiver in the household, but we can equally imagine that a grand-parent moves in precisely to take over child care when the parent decides to enter the labor market. But omitting these variables would create a bias, if they correlate with the unemployment status and decision of the couple to participate in the labor market. For instance, in the case of an unemployment shock to the main provider, the labor income of other adult members is more than possibly a substitute for female participation. Failing to control for household labor income would result in a downward biased estimation of γ . I therefore choose to control for the full set of covariates. $\epsilon_{h(r),t}$ is clustered by household, since unemployment, my variable of interest, varies at the household level.⁹

Interpreting γ — Even so, γ still need not be interpretable as the causal effect of male job loss on female participation. In general, household members — notably, life partners — make their labor supply decision jointly. This has been shown in a variety of empirical applications. For instance, labor supply interdependencies between spouses have been identified in the context of an exogenous workweek reduction reform (Goux et al., 2014). The participation of female partners over the lifecycle happens to be considerably influenced by joint taxation schemes (Groneck and Wallenius, 2021). The retirement eligibility of a partner causally impacts the decision of the other partner to retire (Lalive and Parrotta, 2017). Even more directly related to this paper, the literature provides theoretical foundations and empirical support for the joint job-search behavior of couples. For Guler et al. (2012), a job offer will quickly be accepted by a dual-searcher couple (in relation to a single individual), because job search options are still open for the second spouse. The acceptance of a job by a searching spouse may then trigger the employed spouse to quit one job for a better one. This gives rise to a 'work-quit-search-work' dynamic within the couple. In my context, a male partner may encourage his wife to enter employment, before he quits one job and starts searching for a better one.

From an econometric point of view, this boils down to a reverse causality issue, where partners' decisions mutually influence each other. Since female participation can cause male unemployment, such lifetime arrangements should bias γ upward.¹⁰ To address this possibility, I now turn to an instrumental variable strategy, where the unexpected component of the industry-by-occupation shock to male employment during the 2002 economic crisis serves as an instrument to male job loss.

Instrumental variable strategy — My instrumental variable relies on a quasi-experiment, namely, the sudden collapse of the convertibility regime in January 2002 in Argentina. Section 2 offered key background information regarding the convertibility law, and showed how a series of events built up to trigger the abrupt end of the convertibility regime. In particular, I provided a body of evidence pointing to the fact that the end of convertibility was largely unexpected by the public before autumn 2001. In addition, I now show that this sudden event changed the exposure of individuals to a job loss, to a degree that they could not have predicted from the employment trend of the preceding mild recession. I measure this exposure intensity at the industry-by-occupation level, with the aim of using it as an instrument for the actual job loss experienced by male partners between 2001 and 2002. Finally, I discuss the identifying assumptions underlying the use of this instrument.

Industry-by-occupation exposure – To capture the unexpected component of the industry-by-occupation male exposure to unemployment following the sudden collapse of convertibility, I rely on the full sample of men aged 16-64 between 1996 and 2002.¹¹ $E_j^{i,o}$ is a dummy equal to 1 if individual j is employed in industry i and occupation o, 0 otherwise.¹² For each industry-occupation pair, I estimate the following specification:

$$E_{j(r),t}^{i,o} = a^{i,o}Trend_t + b^{i,o}Post_t + \phi_r^{i,o} + \epsilon_{j(r),t}^{i,o}$$

$$\tag{2}$$

where $\phi_r^{i,o}$ is a set of province fixed effects, controlling for differences in industrial composition across labor markets, *Trend* is a linear time trend, and *Post* is a dummy equal to 1 if individual *j* is observed after the collapse of convertibility and therefore either in May, or October 2002; it equals 0 otherwise. $a^{i,o}$ thus captures the structural evolution of male employment in the industry-occupation pair *i*, *o*, as anticipated by households. By contrast, $b^{i,o}$ measures how male employment in the industry-occupation pair i, o reacts to the sudden end of convertibility, and serves as an instrument for a male partner's job loss.

Figures B1 and B2 in the Appendix display point estimates $\hat{a}^{i,o}$ and $\hat{b}^{i,o}$ for all industryoccupation pairs *i*, *o*, along with their 90% confidence interval. For instance, $\hat{b}^{12,3}$ (operators in the transportation sector) is -0.008, so the probability for men aged 16-64 of being employed in this particular industry-occupation relative to being employed in another industry, or unemployed, or inactive, decreases by 0.8 percentage points with the collapse of convertibility. In the years before the collapse, this particular industry-occupation pair had shown signs of relative expansion, as suggested by the positive $\hat{a}^{12,3}$ point estimate. As visible on Figure B1, before the sudden end of convertibility, men's employment in some sectors was already on a relatively upward trend – with the example of clerical workers in the private and public service sectors – or a downward trend – with the example of technicians and operators in the manufacturing sectors. As the convertibility regime collapsed in January 2002, one in every four industry-occupation pairs (weighting for 40%of total male employment) experienced a significantly negative unexpected employment shock. Employment was hardest hit in the construction and wholesale trade. This was expected, for these industries were the one affected by the greatest shocks to product demand: a 33.4% GDP recession in construction in 2002, and one of 18.4% in wholesale and retail trade (McKenzie, 2004). Other industries were affected as the crisis spread. The only industries where the probability of employment increased in 2002 relative to other industries and unemployment or inactivity were industries where the rate of selfemployment was high (such as repair for technicians, or domestic services for operators), or sectors where labor demand was supported by the State (such as local social services, or public administration for the least qualified).

Instrumental variable model – $\hat{b}^{i,o}$ measures the severeness of the shock to male employment in different industry-occupation cells upon the collapse of the convertibility regime. Therefore, my instrumental variable $Intensity_{i,o,t}^m$ takes the value of 0 for all couples observed before January 2002, and $-b^{i,o}$ for the May and October 2002 waves.¹³

The two-stage empirical specification is given by:

$$Unemployment_{h(i,o,r),t}^{m} = \pi Intensity_{i,o,t}^{m} + \eta'_{i,o} + X_{h,t}\beta' + c'_{h} + \mu'_{r,t} + \epsilon'_{h(i,o,r),t}$$

$$Y_{h(i,o,r),t}^{f} = \gamma Unemployment_{h,t}^{m} + \eta_{i,o} + X_{h,t}\beta + c_{h} + \mu_{r,t} + \epsilon_{h(i,o,r),t}$$
(3)

where $Intensity_{i,o}^m$ is the instrument defined above, and $\widehat{Unemployment^m}_{h,t}$ is the predicted value for male unemployment, as given by the first stage equation. To the covariates $X_{h,t}$ and fixed effects c_h , $\mu_{r,t}$ described above, I add a set of industry-occupation fixed effects $\eta_{i,o}$, identified in men changing their industry-occupation cell over time, because omitting these fixed effects could lead to biased estimates. For instance, during the six month window between October 2001 and May 2002, women may have entered the labor market, and men may have found a job in an other, less exposed industry. Failing to control for the switch would lead to a downward bias of the estimate, because a woman's participation would be mistakenly associated with a relatively low industry-occupation exposure for her partner. $\epsilon_{h(i,o,r),t}$ is clustered by industry-occupation pair, because the instrument varies at the industry-occupation level.

Validity of the exclusion restriction – For the empirical strategy exposed above to be valid, the exposure of men to unemployment should affect women's participation only through the channel of their partner's job loss, and not through some other channel. In particular, the exclusion restriction loses its validity if their partners' exposures to a job loss are correlated.

An obvious case of correlation arises when both partners work in the same industry. Figure B3 in the Appendix pictures the distribution of the partners' industry combinations for dual-earner partners in October 2001. Overall, partners are more likely to work in a different industry than in the same one, and, for most industries, the conditional distribution of industries which employ females is rather uniform. This confirms the view that industries are gender segregated, a stylized fact also documented in developed economies (Verdugo and Allègre, 2020). Two instances stand out, namely, retail trade, reflecting the existence of family businesses, and to a lesser extent the public sectors (administration, education, and health), reflecting assortative matching. In the discussion, I show that my results are robust to the exclusion of couples where male partners work in retail trade, or in these public sectors. More broadly, I also show that my results are robust to the exclusion of same industry couples, defined as couples reporting working in the same industry (or being unemployed after a job in the same industry) at least once before and after the end of convertibility.

Exposure to the end of convertibility can be correlated across partners, whether or not they work in the same industry. To check whether the exposure to the convertibility shock is correlated within couples, I start by estimating again Equation (2) for all industry-occupation pairs in the subsample of *women* aged 16-64. This allows me to assign to each dual-earner couple observed in October 2001 in a given industry and occupation combination its own combination of future shocks. Figure B4 plots all these combinations. The size of the circle represents the weight of a particular combination of male-female shocks over the total number of existing combinations. Most combinations are weighted below 0.1%, but a few combinations are weighted more than 1%: clerical workers in public administration with technicians in education (1.12%), couples of technicians in education (1.20%), couples of clerical workers in public administration (1.55%), couples with an elementary occupation in retail trade (2.88%), and construction operators in couple with unqualified domestic workers (3.3%). As reflected by the solid line in Figure B4, a simple weighted regression of the future shock experienced by woman on her partner's future shock suggests that shocks are slightly positively correlated within households: men working in the industry-occupation pairs most affected by the end of convertibility tend to be the partners of women who are also slightly more exposed than other women. However, this effect is essentially driven by the joint negative shock experienced by construction operators whose partners are unqualified domestic workers (represented by what is highlighted in black at the bottom left-hand corner of Figure B4). When I drop these dual-earner couples, the positive correlation disappears, as reflected by the dashed line also incorporated on Figure B4. In Section 6, I show that my core results are robust to the exclusion of couples where the men are construction workers.

5 Results

I now turn to the presentation and discussion of the results obtained by estimating the specifications outlined in Section 4.

Baseline results – Column 1 of Table 2 presents the results of the baseline specification as described above. Columns 2 and 3 further investigate the extensive margin of participation, and look at female employment and full-time employment (defined as working 20 hours per week or more). Columns 4 and 5 document the responses at the intensive margin on the subsample of working women.¹⁴

[Table 2 about here]

As seen in Column 1, women are 6 percentage points more likely to enter the labor market when their partner loses his job. This is a 12% increase from the 50.3% baseline participation indicated in the bottom part of the table. As displayed in Columns 2 and 3, for these women, employment increases by 4 percentage points and full-time employment (over 20 hours per week) by 2 percentage points. So out of three new female entrants in response to their partner's job loss, two actually find a job, and one even starts working full-time within 3-6 months.¹⁵

Looking at the correlation with other covariates, women's labor market decisions depend very little on changes in the household composition. Notably, they do not respond to changes in the number of children.¹⁶ In line with expectations, female participation is a substitute for most other non-labor sources, such as pensions, unemployment benefits, or remittances. Such income is however a complement to other sources of labor income. This may be because the family business is expanding, or because the decision to participate also concerns other family members beyond the female partner. As shown in Appendix Table A5, the results displayed in Table 2 are robust to the exclusion of these covariates. This is important because it suggests that time-varying unobserved heterogeneity is unlikely to bias the results (Oster, 2019).

Finally, I turn to the intensive margin. Columns 4 and 5 indicate how underemployment and weekly hours vary with the job loss of a male partner. As can be seen in Column 4, the partner's job loss is associated with a 25% increase in the probability that a woman wishes to work more. However, Column 5 shows that it is not correlated with the actual number of hours worked.

The suggestive evidence presented in Table 2 points towards a sizable correlation between unemployment and the labor market outcomes of life partners. When I estimate a comparable specification on the pooled cross-sectional data, excluding household fixed effects (not reported here), γ is two to three times higher, even when I control for the richest possible set of covariates. The panel dimension is thus crucially important for avoiding a large upward omitted variable bias. The period-by-province fixed effects account for the combined effect of local labor market opportunities for both partners, so γ captures the effect of male unemployment, net of the changing economic environment. Therefore, I consider the results summarized in Table 2 to be informative *per se*, and an important step in underpinning a causal effect of a man's job loss on the labor outcomes of his female partner.

However, as discussed in Section 4, the results still need not be interpretable as the causal effect of male job loss on female participation, because partners' decisions mutually influence each other, generating a reverse causality problem. This is why I now turn an instrumental variable strategy, where I estimate women's participation and employment reactions to their partners' *exogenous* job loss (through the unanticipated, heterogeneous shock to male employment, fueled by the end of convertibility).

IV results – Consistently with the empirical strategy exposed in Section 4, I estimate the set of equations (3) on the subsample of partners observed at least once in or before October 2001, and once after the convertibility collapses, i.e. in May and/or October 2002. My sample consists of 20,071 observations of 6,268 unique households between October 2000 and October 2002. ¹⁷

[Table 3 about here]

Table 3 displays the first stage estimates for the extensive and intensive margin estimations. As expected, the correlation between the intensity of the shock to male employment and mens' reported unemployment is significant and positive. As indicated in the bottom part of Table 3, the average shock is 0.001, with a standard deviation of 0.005. In concrete terms, a one standard deviation change in the shock intensity is associated with a $(5.611 \times 0.005 =)$ 2.8 percentage point increase in the likelihood for male partners to be unemployed. At the bottom, I report the KP F statistic as a test for weak instruments. At the extensive margin, the statistic is far above the critical values provided by Stock and Yogo (2005). When I implement the first stage on the subsample of dual-earner couples, the statistic, albeit much smaller, also lies in an acceptable range.

[Table 4 about here]

Table 4 presents the results for the two-stage estimation. Compared to the average level of female participation and employment when the partner is employed (47.5% and 42.6%, respectively), the probability of participating and being employed when the partner unexpectedly loses his or her job is multiplied by two.¹⁸ However, the probability of finding a full-time job does not increase with a partner's displacement. Working full-time may require anticipatory action from women who are also caregivers in their household. In addition, the low labor demand may also condemn women to underemployment. This latter explanation is supported by the results on the intensive margin. Working women do not work more hours after their partner loses his job, but they vehemently wish that they could.

6 Discussion

In this section, I discuss the robustness of my core results. I also investigate the mechanisms that may be driving these results. **Robustness** – First, I provide additional supporting evidence of the validity of the exclusion restriction. I start by testing the sensitivity of my results to the exclusion of different male industries. To do so, I estimate again the set of equations (3), excluding one industry at a time. Figure C1 shows the coefficients obtained for these first stage and second stage estimations. Sub-figure C1a reports the female participation response to male unemployment on these different subsamples. Sub-figure C1b displays the corresponding first stage male unemployment response to the intensity of the convertibility shock. As can be seen in Figure C1, my main results are not driven by any particular male industry. Note that excluding construction introduces some noise, because this industry is weighted as of October 2001 for 11% of the employed men, and 40% of the unemployed men aged 16-64. Still, the sign and magnitude of the coefficients do not change upon the exclusion of this industry.

More broadly, I also test the sensitivity of the results to the presence of couples working in the same industry. To the extent that shocks affecting women and men in the same industry are likely to be positively correlated, I expect these women to be more discouraged from participating in the labor market than others. A couple works in the same industry if both partners report working in the same industry (or being unemployed after a job in the same industry) at least once before and after the end of convertibility. These couples represent 6.4% of the total sample, and 14.3% of the sample of couples in which women work. I estimate again the set of equations (3), excluding couples working in the same industry. Table C2 reports the female participation response to male unemployment on this subsample. The estimates are very similar to the main 2SLS coefficients presented in Table 4. They tend to be higher in magnitude, supporting the idea that in the presence of positively correlated shocks, my main results likely measure the lower bound of the true parameter values. Indeed, in this case, a woman in a relationship with an exposed man would also have a higher probability than other women of being discouraged from entering the labor market.

Then I check whether the shock intensity captured by my instrumental variable in 2002 is statistically related to a pre-trend in male unemployment before the convertibility collapse in 2002. To this end, I estimate again the first stage of Equation (3), augmented with interaction terms between year dummies and the 2002 shock intensity, on couples aged 18-64, observed between 1996 and 2002:

$$Unemployment_{h(i,o,r),t}^{m} = \sum_{y=1996}^{2002} \pi^{y} \left[Intensity_{i,o,2002}^{m} \times \mathbb{1}(\text{Year} = y) \right] + \eta'_{i,o} + X_{h,t} \beta' + c'_{h} + \mu'_{r,t} + \epsilon'_{h(i,o,r),t}$$
(4)

Figure C2 plots the point estimates $\hat{\pi}^y$, together with their 90% confidence interval. Taking 1996 as a reference point, I find that the probability of being unemployed does not increase faster in the industry-occupation pairs that will be most hit in 2002. From 1997 to 2000, male unemployment evolves as rapidly, or even less rapidly in these industryoccupation pairs than in the rest of the economy. However, in 2001, the difference in unemployment trends across industry-occupation pairs is positively related to the intensity of the 2002 shock. In other words, taking 1996 as a reference point, unemployment in more exposed industry-occupations was already growing at a slightly faster rate a year before the actual end of convertibility.

I argue that this divergence does not pose a serious threat to identification. First, in absolute terms, the magnitude $\hat{\pi}^{2001}$ lies in the same range as $\hat{\pi}^{1997} - \hat{\pi}^{2000}$, and is comparable to $\hat{\pi}^{1998}$. Then, by contrast, the divergence in unemployment once the convertibility collapses, measured by $\hat{\pi}^{2002}$, is five times higher than the unemployment divergence captured by $\hat{\pi}^{2001}$. In other words, the probability of being unemployed decreases and then increases slightly if one works in an industry more exposed to the 2002 crisis, without these differences being significant from one year to another. It is only between 2001 and 2002 that a significant and positive divergence appears. Therefore, $\hat{\pi}^{2002}$ stands out as the only wide divergence in unemployment trends.

I close this section by providing some evidence on another potential threat to validity, namely non-random panel attrition. I consider non-random panel attrition to be a potential issue in the present research context, for two reasons. First, by design, the survey follows dwellings rather than households, so couples moving out automatically generate panel attrition. It has been documented that changes in labor market opportunities are a key driver of mobility, so the probability of attrition could be positively correlated with the convertibility shock. Then, in May 2002, the national statistical institute recorded threats and assaults on EPH interviewers. These violent acts caused attrition, notably in the Great Buenos Aires area (INDEC, 2002), and are probably correlated with particularly bad labor market outcomes.

To what extent does the likelihood of dropping out of the sample depend on the shock intensity experienced by male partners in 2002? To answer this question, I first use the main sample described in Section 5, and estimate the following specification:

$$Attrit_{h(i,o,r),t} = \gamma Intensity_{i,o,2002}^m + X_{h,t}\boldsymbol{\beta} + c_h + \mu_{r,t} + \epsilon_{h(i,o,r),t}$$
(5)

where $Attrit_{h(i,o,r),t}$ is equal to 1 if household h living in region r in period t drops out of the survey prematurely at t + 1, 0 otherwise, $Intensity_{i,o,2002}^{m}$ is the shock intensity measure calculated in Section 4 for the industry-occupation pair i, o of the male partner, c_h are household fixed effects, $\mu_{r,t}$ are province period fixed effects, and $X_{h,t}$ are the same covariates as in Equation (1). I thus identify γ on variations in the shock intensity within households, since a male partner may change industry and/or occupation. Second, for external validity, I test whether the probability of attrition varies *across* households, depending on their exposure to the 2002 shock. To this end, I estimate again Equation (5) using pooled OLS, where I replace household fixed effects with the basic demographic characteristics of both partners, i.e. their age and their level of education using four stages (none, primary, secondary and tertiary education). Columns (1) and (2) of Appendix Table C3 show that the likelihood of dropping out of the survey is not related to an increased exposure to the convertibility crisis within households, nor to a difference in exposure *across* households.¹⁹ Finally, I take the attrition issue upstream. Indeed, my sample restriction imposes the condition that partners should be observed at least once in October 2001 or before, and once after. If the likelihood of being observed after October 2001 depends on the intensity of the shock, my results may be driven by a composition effect. Columns (3) and (4) replicate the estimations of Columns (1) and (2) on the sample of households surveyed between October 2000 and October 2002 before I imposed the restriction. I do not find any evidence that households about to be hit by a more intense shock have a greater tendency to leave the survey.

Mechanisms – I now explore the mechanisms behind my main results. I start by investigating the role played by a state-wide workfare policy, the program *Jefes y jefas*, in sustaining the demand for labor. Then I study whether the causal link running from male unemployment to female participation is driven more by job destruction, job creation, or equally by both. Finally, I extend my research question to other male labor outcomes, and provide complementary evidence on a woman's labor market response to a decline in her partner's monthly labor income.

As exposed in Section 2, the workfare program *Jefes y jefas* was first introduced in January 2002, and reached more than 2 million beneficiaries within a few months. 20 weekly working hours were required as a counterpart for the 150AR\$. In practice, in most cases, the local municipalities in charge of the program proposed part-time, unskilled positions in the public sector that were particularly attractive to women. Indeed, in their evaluation of the program, Galasso and Ravallion (2004) report that two thirds of the beneficiaries were women. Anecdotal evidence suggests that women preferred to enroll themselves on the program as unemployed household heads, while their unemployed partners kept searching for higher paid jobs. To what extent did the workfare program facilitate the response of female participation to male unemployment? To investigate this question, I estimate again the set of equations (3), adding female enrollment into *Jefes y jefas* as a binary control variable.²⁰

[Table 5 about here]

Table 5 displays the second stage results (first stage results are shown in Appendix Table C4). As expected, female enrollment closely correlates with both female participation and employment. Conditional on the program, the causal link from male unemployment to female participation and employment is still positive and significant, albeit divided by two.²¹ I conclude that the new jobs created by *Jefes y jefas* account only for roughly half of the increase in participation and in employment caused by the sudden displacement of male partners.²²

Another interesting question is whether the causal positive link from male unemployment to female participation is driven by men *entering* or *exiting* unemployment. Indeed, in the context of a deep economic crisis, one could as well imagine that a discouraged woman would withdraw from the labor market as soon as her partner found a job in a relatively protected sector. To decide between these alternative interpretations, I estimate the following first-difference specification:

$$\Delta Y_{h(r),t}^{f} = \gamma^{entry} Entry_{h,t}^{m} + \gamma^{exit} Exit_{h,t}^{m} + \Delta X_{h,t} \boldsymbol{\beta} + \mu_{r,t} + \Delta \epsilon_{h(r),t}$$
(6)

where $Entry_{h,t}^m$ is a binary variable equal to 1 if the male partner *enters* unemployment $(\Delta Unemployment_{h,t}^m = 1)$, 0 otherwise, and $Exit_{h,t}^m$ is a binary variable equal to 1 if he *exits* unemployment and finds a job $(\Delta Unemployment_{h,t}^m = -1)$, 0 otherwise.²³ As summarized in Table 6, the causal link running from male unemployment to female participation is exclusively driven by the *loss* of a male job.²⁴

[Table 6 about here]

Besides the rise in unemployment, the sudden resurgence of inflation in 2002^{25} had important consequences on other labor market outcomes, notably wages. McKenzie (2004) documents that 78% of the surveyed households experienced real income declines in 2002, and 63% suffered a real income fall of 20% or more. Given the deindexing of labor contracts during the 1990s, I conjecture that the magnitude of the fall in real wages in a given industry-occupation pair should be related to the variation in the demand for labor in this particular industry-occupation. In other words, the shock intensity on male employment at the industry-occupation level can be used as an exogenous source of variation

for male partners' real labor income. To capture the causal effect of male labor income on female participation and employment, I therefore estimate the set of equations (3) on the sub-sample of couples with an employed male partner, where the variable of interest is not their unemployment status, but the logarithm of their real monthly labor income.

[Table 7 about here]

The first stage results in Table 7 confirm the negative impact of job destruction on wages. A one standard deviation increase in the intensity of the shock to male employment results in a 5.6% decrease in male labor income.²⁶ The second stage results show that female participation and employment respond positively to a decline in real male labor income. The bottom part of Table 7 indicates that the average male labor income decreased by 33% between 2000-2001 and 2002, implying an 8 (resp. 7) percentage points increase in participation (resp. employment).²⁷ I further unpack this result, and investigate how this response varies across the pre-crisis income distribution. To this end, I interact my endogenous variable and my instrument with five binary variables, indicative of couples' household income quintile (per capita) before 2002. An effect concentrated on the poorest households would suggest that female participation is driven by necessity. By contrast, a homogeneous effect would rather stand in support of a ratchet effect, where households seek to maintain their pre-crisis standard of living.²⁸ The right panel of Table 7 supports the necessity motive over the ratchet effect. The participation and employment responses stem exclusively from the first two quintiles of the household income distribution.

7 Dynamics of Female Participation

So far, I have documented the contemporaneous female participation and employment responses to male unemployment and income loss. I now propose two complementary approaches for studying the dynamics of female participation.

Delayed response – First, I rely on the panel dimension of my data, and investigate possible delays in female participation or employment response to male unemployment. In Turkey, using quarterly SILC employment data between 2007 and 2010, Ayhan (2018) finds that the positive female labor supply response appears with a lag of one quarter, and operates for two quarters only. In a similar vein, I estimate again Equation (1), adding lagged male unemployment as a control variable. In this way, I can capture whether female participation also increases with some delay, holding the contemporaneous response to male unemployment constant.

Table 8 reports the results. Note that because of the lagged term, the sample size decreased by 60%. Hence, for comparison purposes, I also report the effect measured without lagged male unemployment. Table 8 confirms that the contemporaneous participation and employment responses are large in size. In addition, I find that both female outcomes also react to male unemployment, with some delay.²⁹ Note that the contemporaneous response to male unemployment is higher for female participation (+6.3 percentage points) than for female employment (+4.2 percentage points). However, the delayed response is of similar magnitude (1.7 to 1.8 percentage points). This illustrates the fact that finding a job requires some delay: unlike participation, employment is limited by the demand for labor.

[Table 8 about here]

Medium run response – When studying the dynamics of female labor supply, another important question is whether the increase in female participation during a crisis can be long-lasting. As soon as 2003, and until 2008, Argentina returned to a sustained 8-9% growth rate each year. With the recovery, did new entrants withdraw from the labor market, or did they rather remain active, possibly revealing an attachment to the labor market?

To the best of my knowledge, the literature on developing economies is silent in this regard. In this paper, I am also constrained by data availability. First, in the *EPH*, households are followed over four waves, that is, for 18 months in total. This time span is insufficient to document any medium-term effect on female participation or employment. Even more critical, as explained in Section 3, the *EPH* survey methodology evolved dramatically in 2003, creating a break in the panel. Therefore, I cannot rely on the panel dimension of the data to measure how female labor market outcomes evolved with the economic recovery.

In what follows, I present descriptive evidence on female participation and employment responses in the medium term. To this end, I rely on the main sample presented in Section 3. First, I assign to each male partner observed between 1996 and 2007 a correspondingly intense shock for 2002, according to his industry-occupation pair observed in t. Then, to capture the medium term participation and employment responses, I interact my measure of exposure to the convertibility collapse with the actual period of observation t. The reduced form specification in cross-section is given by:

$$Y_{h(i,o,r),t}^{f} = \sum_{t=1996q1}^{2007q4} a^{t} \left[Intensity_{i,o,2002}^{m} \times \mathbb{1}(\text{Period} = t) \right] + X_{h,t}^{f} a^{f} + X_{h,t}^{m} a^{m} + X_{h,t} a + \phi_{r,t} + u_{h(i,o,r),t} a^{r} + A_{h,t}^{m} a^{m} + X_{h,t} a + \phi_{r,t} + u_{h(i,o,r),t} a^{r} + A_{h,t}^{m} a^{m} + X_{h,t} a^{r} + A_{h,t}^{m} a^{m} + A_{h,t} a^{r} + A_{h,t}^{m} a^{m} + A_{h,t}^{m} a^{m}$$

where $Y_{h(i,o,r),t}^{f}$ stands for both female participation and employment status, $Intensity_{i,o,2002}^{m}$ is the shock intensity assigned to the male partner in household h given his current industry and occupation (i, o) at period t, $\mathbb{1}(\text{Period} = t)$ is a dummy variable equal to 1 if the household is observed in t, 0 otherwise. $X^{m,f}$ are matrices with both partners' age (in years) and completed level of education (in four stages, as detailed above), $X_{h,t}$ are the usual time-varying household controls, $\phi_{r,t}$ are the province-by-period fixed effects, notably controlling for the evolution of the local labor market opportunities for men and women in time. $u_{h(i,o,r),t}$ is two-way clustered by household and by male industryoccupation pairs (because $Intensity_{i,o,2002}^{m}$ varies at the industry-occupation level).

Figure 1 plots \hat{a}^t . The first point estimate is the base effect of $Intensity_{i,o,2002}^m$ on female participation (1a) or employment (1b) in May 1996. At each period t, \hat{a}^t captures the differential trend in women's participation, according to their male partners' degree of exposure to the 2002 shock, with respect to May 1996. When t < 2002q2, \hat{a}^t is generally not significantly different from 0, confirming the absence of anticipation already discussed in the robustness section. \hat{a}^{2002q2} and \hat{a}^{2002q4} are positive, and measure the contemporaneous participation response to the shock.³⁰ In 2003 and beyond, a^t coefficients capture whether the female participation response lasts in the context of a very rapid economic recovery. I find this response to persist at least until 2005, and possibly even until the end of 2007, albeit smaller in size.

[Figure 1 about here / Figure 1a and 1b are side by side.]

Discussion – An underlying hypothesis behind my interpretation of a^t is that I can rely on male partners' *contemporaneous* industry and occupation to infer the intensity of their exposure to the convertibility crisis in 2002. In other words, I suppose that the male partners observed in t are either working in exactly the same industry-occupation pair as they were when the convertibility collapsed in 2002, or in a very similarly exposed one. Although I cannot test this hypothesis directly, I find that the correlation between lagged and present male exposure in the panel is high and positive (0.66). Even so, a positive \hat{a}^t beyond 2003 need not be interpretable as reflecting an *attachment* to the labor market. Women may simply increase their participation over a longer period of time in order to compensate for the deep income loss incurred in 2002. To shed additional light on the eventual compensation mechanism, I test whether the female participation response in 2003 and beyond is higher for women partnered with lower-income men, holding the 2002 shock intensity constant. If the persistence is due to compensation, rather than attachment, then a woman's participation should be higher when her partner's labor income in t is low, and this negative relation should increase over time, because only women partnered with the lowest earners would remain active. Adding to specification (7) a full set of interactions between the 1(Period = t) dummies, $Intensity^m_{i,o,2002}$, and the real log labor income reported by the male partner, I did not find any evidence in support of compensation.

8 Conclusion

Employment is one of the most important elements in women's empowerment, and ultimately in economic development. In this paper, I find that necessity is still an important motive for female participation, even in the context of an upper-middle-income country like Argentina, where fertility is declining and female education is expanding. Between 1996 and 2007, the correlation between partners' labor outcomes is close: when her partner becomes unemployed, a woman is 6 percentage points more likely than before to enter the labor market. Decomposing this effect, I find that one third of new female entrants find a full-time job, one third settles for a part-time job, and the remainder are unemployed. Then, to properly measure the causal participation response to a negative, unanticipated shock on male employment, I rely on the sudden repeal of the convertibility law in January 2002, which had been the foundation of the Argentine economic system since 1991. I demonstrate that this unexpected legal decision caused a specific change in male employment relative to the underlying employment trend, differing across the 22×4 industry-occupation pairs, which I used as an exogenous source of variation to instrument the actual job loss experienced by male partners around January 2002. In this context, I find that the female labor supply response is very high at the extensive margin: when unemployment hits her partner, a woman is twice as likely to enter the labor market and work. The rapid expansion of a poverty alleviation workfare program in May 2002 contributes to a relaxation of the constraints on the labor demand, and accounts for half of the effect. At the intensive margin, women do not actually increase their working time, despite the fact that they vociferously declare that they would rather work longer hours, signaling either institutional rigidity in working hours, or constraints on the demand side. Using a similar approach for couples where the male partner remains employed, I find that the female labor supply also responds positively to a negative shock on earnings. I do not find evidence that this negative income effect fades out over time

between 1996 and 2007.

The other important contribution of this paper is to shed light on the potential impact of such a labor supply response in the medium term. First, this response is for the most part contemporaneous to the negative shock, but I also find a significant delayed reaction, suggesting that adjustments take time. Then, contrary to expectations, while women *enter* the labor market as their partner loses his job, I find no symmetric evidence that they *withdraw* from the labor market upon their partner's re-employment. Finally, consistently with this latter finding, the repeal of the convertibility law appears to have had persistent impacts on female participation. Therefore, even responses to temporary negative shocks probably have lasting consequences for women's labor force participation.

To conclude, the most recent contributions in the literature have made very important progress in the identification of the structural barriers to female employment, as well as in the design of innovative policies to address them (see Jayachandran (2020) for a review). With this country case study, I provide causal evidence on another mechanism highlighted in the literature, where a woman's participation can be explained by necessity, even in countries with intermediate level of income. Cross-country panel studies generally show that women's participation response in times of crisis is moderate or non-existent in upper-middle-income countries (Signorelli et al., 2012; Cho and Newhouse, 2013; Klasen et al., 2021). However, they usually use aggregated data which are by definition not suited to studying shock-coping mechanisms at the intra-household level. Based on microdata, I find that intra-household labor supply strategies exist even when the variation in female participation is small or negative at the aggregate level. The cross-country study proposed by Klasen et al. (2021) relies on microdata, but it cannot identify causal parameters, since there is no quasi-experimental strategy applicable to all countries and years. Here, I find that the causal effect of an unexpected job loss on female participation is stronger than what reduced form correlates suggest.

My paper thus contributes to the broader understanding of female labor force participation by proposing an in-depth exploration of female participation and employment responses in the context of negative shocks, and by shedding light on their role in the dynamics of the female labor force in the longer term. Contributing to the existing literature on participation for the sake of necessity is important, since it also highlights the potential negative consequences of the labor supply on women, such as reduced leisure, or the double burden of caring and work, often overlooked in the literature (Heath and Jayachandran, 2017). This is an important direction for future research.

References

- Andersen, A. L., Jensen, A. S., Johannesen, N., Kreiner, C. T., Leth-Petersen, S., and Sheridan, A. (2021). How Do Households Respond to Job Loss? Lessons from Multiple High-Frequency Data Sets. Technical report, CEPR Discussion Paper No. DP16131.
- Ayhan, S. H. (2018). Married Women's Added Worker Effect during the 2008 Economic Crisis - The Case of Turkey. *Review of Economics of the Household*, 16(3):767–790.
- Bertranou, F. M. and Paz, J. A. (2007). Políticas y programas de protección al desempleo en Argentina. Technical report, Organización Internacional del Trabajo.
- Blau, F. and Kahn, L. (2007). Changes in the Labor Supply Behavior of Married Women: 1980–2000. Journal of Labor Economics, 25(3):393–438.
- Bredtmann, J., Otten, S., and Rulff, C. (2018). Husband's Unemployment and Wife's Labor Supply: The Added Worker Effect across Europe. *ILR Review*, 71(5):1201–1231.
- Cerrutti, M. (2000). Economic Reform, Structural Adjustment and Female Labor Force Participation in Buenos Aires, Argentina. *World Development*, 28(5):879–891.
- Cho, Y. and Newhouse, D. (2013). How did the Great Recession Affect Different Types of Workers? Evidence from 17 Middle-Income Countries. World Development, 41:31–50.
- Corrales, J. (2002). The Politics of Argentina's Meltdown. World Policy Journal, 19(3):29–42.
- Cullen, J. B. and Gruber, J. (2000). Does Unemployment Insurance crowd out Spousal Labor Supply? *Journal of Labor Economics*, 18(3):546–572.
- Daseking, C., Ghosh, A. R., Lane, T. D., and Thomas, A. H. (2005a). Lessons from the Crisis in Argentina, chapter VI Role of the IMF, pages 39–41. International Monetary Fund.
- Daseking, C., Ghosh, A. R., Lane, T. D., and Thomas, A. H. (2005b). Lessons from the Crisis in Argentina, chapter III Downturn and Deepening Depression, pages 39–41. International Monetary Fund.
- Duflo, E. (2012). Women Empowerment and Economic Development. Journal of Economic Literature, 50(4):1051–79.
- Fanelli, J. M. (2002). Growth, Instability and the Convertibility Crisis in Argentina. Cepal Review, 77:25–43.

- Fiszbein, A., Adúriz, I., and Giovagnoli, P. I. (2003). El impacto de la crisis argentina en el bienestar de los hogares. *Revista de la CEPAL*, 79:151–167.
- Frankenberg, E., Smith, J. P., and Thomas, D. (2003). Economic Shocks, Wealth, and Welfare. *Journal of Human Resources*, 38(2):280–321.
- Galasso, E. and Ravallion, M. (2004). Social Protection in a Crisis: Argentina's Plan Jefes y Jefas. The World Bank Economic Review, 18(3):367–399.
- Gasparini, L. and Marchionni, M. (2015). Bridging Gender Gaps? The Rise and Deceleration of Female Labor Force Participation in Latin America. La Plata, Argentina: Centre for Distributive Labor and Social Studies (CEDLAS), Universidad Nacional de La Plata.
- Giuliano, P. (2017). Gender: An Historical Perspective. *NBER Working Papers No.* 23635.
- Goux, D., Maurin, E., and Petrongolo, B. (2014). Worktime Regulations and Spousal Labor Supply. American Economic Review, 104(1):252–76.
- Groneck, M. and Wallenius, J. (2021). It Sucks to Be Single! Marital Status and Redistribution of Social Security. *The Economic Journal*, 131(633):327–371.
- Guler, B., Guvenen, F., and Violante, G. L. (2012). Joint-Search Theory: New Opportunities and New Frictions. Journal of Monetary Economics, 59(4):352–369.
- Halla, M., Schmieder, J., and Weber, A. (2020). Job Displacement, Family Dynamics, and Spousal Labor Supply. American Economic Journal: Applied Economics, 12(4):253– 87.
- Heath, R. and Jayachandran, S. (2017). The Causes and Consequences of Increased Female Education and Labor Force Participation in Developing Countries. In *The* Oxford Handbook of Women and the Economy.
- Illiano, C. (October 3 2001). Argentina asegura que mantendrá plan económico tras elecciones. *Reuters Noticias Latinoamericanas*. Retrieved from Factiva on November 22nd, 2021.
- ILO (2021). ILOSTAT Database. Available at https://ilostat.ilo.org/data/ (accessed June 23d, 2021).
- IMF (2003). The Role of the IMF in Argentina, 1991-2002. Technical report, Issues Paper/Terms of Reference for an Evaluation by the Independent Evaluation Office (IEO).

- INDEC (2000). Utilización de clasificadores de ocupación en la base de datos de la Encuesta Permanente de Hogares.
- INDEC (2001). Clasificador Nacional de Ocupaciónes. Encuesta Permanente de Hogares.
- INDEC (2002). Encuesta Permanente de Hogares Novedades de la Onda Mayo 2002.
- INDEC (2011). Clasificación de actividades económicas para encuestas sociodemográficas del Mercosur.
- Iturriza, A., Bedi, A. S., and Sparrow, R. (2011). Unemployment Assistance and Transition to Employment in Argentina. *Economic Development and Cultural Change*, 59(4):811–837.
- Jayachandran, S. (2020). Social Norms as a Barrier to Women's Employment in Developing Countries. NBER Working Papers No. 27449.
- Juhn, C. and Potter, S. (2007). Is There Still an Added Worker Effect? NBER Working Papers RRC NB07-14.
- Klasen, S. (2019). What Explains Uneven Female Labor Force Participation Levels and Trends in Developing Countries? *The World Bank Research Observer*, 34(2):161–197.
- Klasen, S., Le, T. T. N., Pieters, J., and Santos Silva, M. (2021). What Drives Female Labour Force Participation? Comparable Micro-Level Evidence from Eight Developing and Emerging Economies. *The Journal of Development Studies*, 57(3):1–26.
- Lalive, R. and Parrotta, P. (2017). How does Pension Eligibility Affect Labor Supply in Couples? *Labour Economics*, 46:177–188.
- McKenzie, D. J. (2003). How do Households Cope with Aggregate Shocks? Evidence from the Mexican Peso Crisis. *World Development*, 31(7):1179–1199.
- McKenzie, D. J. (2004). Aggregate Shocks and Urban Labor Market Responses: Evidence from Argentina's Financial Crisis. *Economic Development and Cultural Change*, 52(4):719–758.
- Menon, N. and Van der Meulen Rodgers, Y. (2015). War and Women's Work: Evidence from the Conflict in Nepal. *Journal of Conflict Resolution*, 59(1):51–73.
- Naranjo, M. (October 5 2001). Oficialismo y oposición argentina critican a De la Rúa. *Reuters - Noticias Latinoamericanas.* Retrieved from Factiva on November 22nd, 2021.

- Oster, E. (2019). Unobservable selection and coefficient stability: Theory and evidence. Journal of Business & Economic Statistics, 37(2):187–204.
- Otaloa, J. A. (October 3 2001). Recuperación en depósitos argentinos pese a cautela ahorristas. *Reuters Noticias Latinoamericanas*. Retrieved from Factiva on November 22nd, 2021.
- Outlook, W. E. (1999). October 1999. International Monetary Fund.
- Outlook, W. E. (2001). December 2001, International Monetary Fund.
- Relea, F. (August 23 2001). El FMI exige duras reformas a Argentina a cambio de un nuevo crédito para salvar el país. *El País*. Retrieved from Factiva on November 22nd, 2021.
- Signorelli, M., Choudhry, M., and Marelli, E. (2012). The Impact of Financial Crises on Female Labour. *The European Journal of Development Research*, 24(3):413–433.
- Solon, G., Haider, S. J., and Wooldridge, J. M. (2015). What are we weighting for? Journal of Human resources, 50(2):301–316.
- Stephens, Jr, M. (2002). Worker Displacement and the Added Worker Effect. Journal of Labor Economics, 20(3):504–537.
- Stock, J. H. and Yogo, M. (2005). Chapter 5. Testing for Weak Instruments in Linear IV Regression. In Identification Inference for Econometric Models: Essays in Honor of Thomas Rothenberg, page 80–108. Cambridge: Cambridge University Press.
- Verdugo, G. and Allègre, G. (2020). Labour Force Participation and Job Polarization: Evidence from Europe during the Great Recession. *Labour Economics*, 66(101881).

Notes

- 1. Source: ILO (2021) modeled ILO estimates.
- 2. Note that the information corresponding to the third quarter of 2007 is not available, since the provinces Mar del Plata-Bátan, Bahía Blanca-Cerri, Gran La Plata and the Greater Buenos Aires area were not surveyed due either to administrative reasons, or a strike of the *EPH* personnel.
- 3. This living arrangement represents 64% of the total sample of households. The remaining 36% are one-person households (16%), and single adults living with children or extended family (20%).
- 4. The survey provides sampling weights, which allow for population labor market statistics to be computed in the cross-section. However, these weights vary for the same household across waves, and they do not account for attrition in the panel, so they are not suited to longitudinal analysis. I prefer to use weighted data when I rely on repeated cross-sections to compute descriptive statistics, and my instrumental variable. In turn, following McKenzie (2004) and Solon et al. (2015), I use unweighted data whenever the analysis relies on the panel dimension of the data. In practice, non-random sampling turns out to be inconsequential, as illustrated by Table A4 which displays the *unweighted* demographic characteristics of the main sample. They are very similar to the *weighted* characteristics presented in Table A1.
- 5. Income is adjusted for inflation using the national consumer price index of Argentina provided by the Federal Reserve Bank of St. Louis (Index = 100 in 1995).
- 6. In Argentina, recall that access to unemployment insurance is restricted to formal workers. The line between unemployment and inactivity may thus be blurred if individuals have no particular reason to search for a job when none are available. For this reason, I also estimate the baseline specification with non-employment as my main explanatory variable. NonEmployment^m_{h,t} is a dummy equal to 1 if the partner is either unemployed or inactive, 0 if he is employed. I find very similar results, both in sign and in magnitude (available upon request).
- 7. 12.5% of the male partners in the panel changed employment status $(Unemployment_{h,t}^m)$ at least once, while 85.8% (resp. 1.7%) remained employed (resp. unemployed). More precisely, 35% of them lost their job, 36.6% found a new job, and 28.3% both lost and found a job during the observation period. These unstably employed male partners live in significantly larger households, with more children and other adults. Uneducated and primary school graduates are over-represented (70.6%, as against 52.5% in the stably employed group).
- 8. Provinces are the smallest geographic unit available in the survey. There are 29 provinces in the survey. The largest province is over 100,000 square mile, which may seem large to account for the changing local economic environment. However, in concrete terms, the survey concentrates on one, or sometimes two urban centers per province. Province fixed effects are therefore rather convincing proxies for local urban labor markets.
- 9. The significance of the results displayed in Table 2 is not affected by the use of alternative standard errors, i.e., standard errors clustered at the province level, or two-way clustered at the province and

period levels.

- 10. This positive reverse causality can be counterbalanced by the presence of leisure complementaries if both spouses enjoy spending time together. Alternatively, the upward bias can be amplified if their leisure times are substitutes in their individual utilities.
- 11. May 1996 saw the first available wave at the national level. In 2003, Argentina entered a strong recovery phase.
- 12. There are 22 industries, and 4 types of occupations in the present study. Section B in the Appendix provides additionnal information about the classifications used by the *EPH*, the definitions of industries and occupations; it also presents descriptive statistics on each industry-occupation pair.
- 13. Note that I use the opposite of $\hat{b}^{i,o}$ rather than $\hat{b}^{i,o}$ itself to get a measure of the *intensity* of the shock to male employment. Since I instrument for unemployment, this allows for a more intuitive interpretation of the results: a higher shock to employment in a given industry-occupation pair implies a higher likelihood of being unemployed for men working in this particular industry-occupation pair.
- 14. Appendix Table A3 compares women working to other women in couples, and also compares women continuously working throughout the panel to women changing employment status over time. As expected, working women are more educated and have fewer children. Interestingly, women with interrupted employment are very similar to inactive/unemployed women in terms of education and household composition.
- 15. Waves are spaced 6 months apart in the *EPH* (1996-2002), and 3 months apart in the *EPH continua* (2003-2007).
- 16. Note that the time span between consecutive observations in the panel is only three to six months, so this apparent absence of correlation may in fact be explained by anticipated withdrawals from the labor market.
- 17. I replicate the baseline specification of Equation (1) on this subsample, and get results that are very similar to Table 2, displayed in Appendix Table C1. In Section 6, I also discuss the possibility that my results are driven by a composition effect due to non-random attrition.
- 18. Indeed, in the event of an unexpected job loss of the partner, women's participation and employment increase by 48 to 49 percentage points. The response to the unexpected job loss is greater than the simple panel estimates suggest (see Table 2 above, as well as Table C1 in the Appendix). This difference is explained by several factors. First, the baseline estimates probably suffer from a downward bias. In normal circumstances, some women participate when their husband is employed, but prefer not to work if their husband is unemployed, either because of leisure complementarities, or because it is difficult to violate the gender norm that a man should provide for the household. Second, the effect measured by instrumental variable is a local effect. It is quite plausible that men whose employment is particularly sensitive to the intensity of the crisis (i.e. men who are unemployed due to the consequences of the crisis, but who would otherwise have kept their jobs) are also the men whose wives are particularly inclined to

react and enter the labor market.

- 19. The coefficient γ is very small: in column (1), a one standard deviation change in the shock intensity is associated with a (0.119 × 0.005 =) 0.06 percentage point increase in the likelihood of dropping out prematurely from the survey.
- 20. The national statistical institute collected data on *Jefes y jefas* beneficiaries in a separate survey, starting in October 2002. I merge this data with my sample, and I use information on the date of entry in the program to generate retrospective information on the beneficiaries in May 2002. I find that 1.9% of female partners (resp. 9.9%) were beneficiaries in May (resp. October) 2002. Note that the program was mostly extended after April 2002; hence the difference. In October 2002, I am also able to document that 83.5% of the female beneficiaries actually complied with the working requirements.
- 21. In the context of the rapid scale-up of Jefes y jefas, one could imagine that the positive participation response to male unemployment is *indirectly* driven by the program. Inactive, non-beneficiary women with displaced partners would file for unemployment, with the sole objective of eventually enrolling on Jefes y jefas. Under this scenario, however, I should find that female employment does not respond to male unemployment. The results in Table 5 contradict this scenario; in fact, male displacement generates both a participation response and an employment response.
- 22. Note that one third of the beneficiaries of this workfare program were unemployed *men*. In the labor force survey, these beneficiaries were classified as employed, which lowered the unemployment rate between the first and second half of 2002. In my sample, the unemployment rate was 16.5% in May 2002 and 12.5% in October 2002. If I consider the beneficiaries as unemployed, these rates increase to 20.2% and 18% respectively. As a robustness check, I estimate again Equation (1), as well as the set of equations (3), where $Unemployment_{h,t}^m$ is now a dummy equal to one if the male partner m is unemployed, or employed under the Jefes y jefas unemployment assistance program. The results are essentially unaffected by this change in the definition of male unemployment (results available upon request).
- 23. In other words, $\Delta Unemployment_{h,t}^m$ is a linear combination of $Exit_{h,t}^m$ and $Entry_{h,t}^m$: $\Delta Unemployment_{h,t}^m = Exit_{h,t}^m Entry_{h,t}^m$. Note that I do not instrument for $Entry_{h,t}^m$ and $Exit_{h,t}^m$, because I have a single instrument.
- 24. Interestingly, women work 3 hours more in response to any change in their partner's status irrespective of entry or exit. This is probably why I find no overall effect of male unemployment on female working hours in Tables 2 and C1.
- 25. Fiszbein et al. (2003) report that the consumer price inflation for the Greater Buenos Aires area was 41% for 2002.
- 26. $-11.21 \times 0.005 \times 100 = -5.6.$
- 27. $((461 689)/689) \times 100 \times -0.00246$ (-0.00215) = 0.0814 (0.0711)
- 28. Note that wealthier households may adopt other strategies, involving savings, or loans, to smooth their

consumption. Also, in the presence of assortative mating, women from wealthier households who do not participate in the labor market despite high earnings potential may have particularly strong traditional gender norms that prevent them from entering the labor market, even more so when their partner is unemployed.

- 29. The time span between two waves is a semester between 1996 and 2002, and a quarter for later waves.
- 30. For the sake of comparison, I also estimate the reduced form $Y_{h(i,o,r),t}^f = \rho Intensity_{i,o,2002}^m + \eta_{i,o} + X_{h,t}\beta + c_h + \mu_{r,t} + \epsilon_{h(i,o,r),t}$ on the October 2000-October 2002 sample used in the main result section, where $Y_{h(i,o,r),t}^f$ is female participation. I find that $\hat{\rho} = 2.72$ (results not reported here). This gives additional credibility to the cross-sectional results displayed in Figure 1, as $\hat{\rho}$ lies between $\hat{a}^{2002q2} = 2.62$ and $\hat{a}^{2002q4} = 5.01$.

	1996-1999	2000-2002	2003-2007
Female labor outcomes			
Participating	0.47	0.49	0.58
	(0.50)	(0.50)	(0.49)
Working	0.41	0.42	0.52
	(0.49)	(0.49)	(0.50)
Working for 20 hours and more	0.28	0.28	0.32
	(0.45)	(0.45)	(0.47)
Observations	89,475	53,751	100,014
Conditionnal on working			
Underemployed	0.31	0.34	0.22
	(0.46)	(0.48)	(0.41)
Hours worked per week	33.53	32.33	29.97
	(19.66)	(19.25)	(19.28)
Monthly wage	516.09	467.72	394.30
	(543.67)	(548.30)	(607.57)
Observations	$36,\!522$	23,186	52,394
Male labor outcomes			
Unemployed	0.09	0.13	0.05
	(0.28)	(0.33)	(0.22)
Observations	89,475	53,751	100,014
Conditionnal on working			
Monthly wage	824.86	710.50	613.67
	(820.85)	(763.42)	(774.12)
Hours worked	48.16	45.96	45.85
	(18.58)	(19.45)	(19.58)
Observations	83,111	47,767	95,832

Table 1: Summary Statistics: Labor Outcomes

Note: The sample used is the main sample described in Section 3, divided into three periods. The core analysis focuses on the period 2000-2002.

	Ext (tensive Margi All Women)	Intensive (Working V	Margin Women)	
	Particip.	Empl.	Full T.	Wish	Hours
Male partner is unemployed	0.0592***	0.0390***	0.0203***	0.0495***	0.142
	(0.004)	(0.004)	(0.004)	(0.008)	(0.239)
Composition					
N children 0-2	0.00535	-0.000534	0.00397	0.00995	1.464***
	(0.004)	(0.004)	(0.003)	(0.008)	(0.267)
N children 3-5	-0.00672	-0.00696*	-0.00332	0.00260	0.612^{**}
	(0.004)	(0.004)	(0.004)	(0.008)	(0.282)
N children 6-15	0.000163	0.00123	-0.00307	0.00239	0.150
	(0.004)	(0.004)	(0.003)	(0.006)	(0.243)
N children 15+ studying	-0.000136	0.00302	0.00156	0.000623	0.479^{**}
	(0.003)	(0.003)	(0.003)	(0.006)	(0.218)
N adults 16-64	-0.00629**	-0.00602**	-0.00456*	-0.000474	0.120
	(0.003)	(0.003)	(0.003)	(0.006)	(0.200)
N adults 65+	0.0232**	0.0187*	0.00782	-0.0327*	0.105
	(0.010)	(0.010)	(0.010)	(0.018)	(0.624)
Household Income	. ,	. ,	. ,	· /	. ,
Other Labor Income	0.00616***	0.00721***	0.00447***	-0.00220**	0.0370
	(0.001)	(0.001)	(0.001)	(0.001)	(0.040)
Pensions	-0.0125***	-0.0128***	-0.00809***	0.00212	-0.195**
	(0.001)	(0.001)	(0.001)	(0.002)	(0.090)
Capital	0.000731	0.000273	-0.000876	-0.00129	-0.00997
-	(0.001)	(0.001)	(0.001)	(0.002)	(0.075)
Unemployment	-0.00451**	-0.0112***	-0.00803***	0.00230	-0.0545
	(0.002)	(0.002)	(0.002)	(0.004)	(0.127)
Remittances	-0.00305*	-0.00876***	· -0.00990***	0.0101***	-0.170
	(0.002)	(0.001)	(0.001)	(0.003)	(0.104)
Other	-0.0116***	-0.0154***	-0.00483***	0.00684***	-0.245***
	(0.001)	(0.001)	(0.001)	(0.002)	(0.067)
Household FE	Yes	Yes	Yes	Yes	Yes
Region \times Time FE	Yes	Yes	Yes	Yes	Yes
\mathbb{R}^2	0.762	0.792	0.755	0.585	0.755
\bar{Y} if Husb. Employed	0.503	0.457	0.308	0.209	32.383
Observations	243,240	243,240	243,240	101,332	101,332
Unique households	81,956	81,956	81,956	36,614	36,614

 Table 2: Baseline Specification

Note: *** denotes significance at 1%, ** at 5%, and * at 10%. S.e. clustered by households are shown in parentheses. Particip. is a female participation dummy, Empl. is a female employment dummy, Full T. is a binary variable equal to 1 if women report working 20 hours or more, Wish is a binary variable equal to 1 if women declare that they are seeking to work longer hours, Hours is a continuous variable measuring the weekly working hours. The sample used is the main sample described in Section 3. In the last two columns, the sample is restricted to couples where the women are working.

	Extensive Margin (All Women)	Intensive Margin (Working Women)
	Male partner is unemployed	Male partner is unemployed
Shock Intensity	5.611***	4.441***
	(0.486)	(0.815)
Household FE	Yes	Yes
Region \times Time FE	Yes	Yes
Industry \times Occupation FE	Yes	Yes
Time-varying controls	Yes	Yes
KP	133.13	29.68
Mean(Intensity)	0.00153	0.00130
SD(Intensity)	0.00499	0.00464
Observations	20,071	7,720
Unique households	6,268	2,642

Table 3: Main Results – First Stage Estimates

Note: *** denotes significance at 1%, ** at 5%, and * at 10%. S.e. clustered by industryoccupation pairs are shown in parentheses. There are two first stage regressions: the first column reports the first stage results for all couples, at the extensive margin of participation and employment, and the second column reports the results on the restricted sample of working women, to document the intensive margin. The sample used is the sample described in Section 5, that is, the sample of couples observed at least once in October 2001 or before, and once after the repeal of convertibility. The time-varying controls are the household composition variables listed in the bottom panel of Table A1, as well as the amounts of household income listed in Table A2. For the sake of completeness, Appendix Table B1 also displays the coefficients of the time-varying controls.

	Exter (A)	nsive Marg ll Women)	Intensive Margin (Working Women)		
	Particip.	Empl.	Full T.	Wish	Hours
Male partner is unemployed	0.484^{***}	0.492^{***}	(0.076)	0.543^{**}	-12.63
	(0.000)	(0.001)	(0.010)	(0.200)	(0.102)
Household FE	Yes	Yes	Yes	Yes	Yes
Region \times Time FE	Yes	Yes	Yes	Yes	Yes
Industry \times Occupation FE	Yes	Yes	Yes	Yes	Yes
Time-varying controls	Yes	Yes	Yes	Yes	Yes
\overline{Y} if Husb. Employed	0.475	0.426	0.296	0.280	33.587
Observations	20,071	20,071	20,071	7,720	7,720
Unique households	6,268	6,268	6,268	$2,\!642$	$2,\!642$

Table 4: Main Results - 2SLS Estimates

Note: *** denotes significance at 1%, ** at 5%, and * at 10%. S.e. clustered by industry-occupation pairs are shown in parentheses. Particip. is a female participation dummy, Empl. is a female employment dummy, Full T. is a binary variable equal to 1 if women report working 20 hours or more, Wish is a binary variable equal to 1 if women declare that they are seeking to work longer hours, Hours is a continuous variable measuring the weekly working hours. The sample used is the sample described in Section 5, that is, the sample of couples observed at least once in October 2001 or before, and once after the repeal of convertibility. In the last two columns, the sample is restricted to couples where women report working at least once before and after January 2002. The time-varying controls are the household composition variables listed in Table A2. For the sake of completeness, Appendix Table B2 also displays the coefficients of the time-varying controls.

	Exter (A	nsive Marg ll Women)	Intensive Margin (Working Women		
	Particip.	Empl.	Full T.	Wish	Hours
Male partner is unemployed	0.239**	0.177^{*}	-0.0839	0.568**	-10.91
	(0.092)	(0.097)	(0.077)	(0.252)	(9.370)
Workfare Beneficiary	0.377^{***}	0.484***	0.0248	-0.0256	-1.795
	(0.022)	(0.022)	(0.032)	(0.089)	(1.408)
Household FE	Yes	Yes	Yes	Yes	Yes
Region \times Time FE	Yes	Yes	Yes	Yes	Yes
Industry \times Occupation FE	Yes	Yes	Yes	Yes	Yes
Time-varying controls	Yes	Yes	Yes	Yes	Yes
\bar{Y} if Husb. Employed	0.475	0.426	0.296	0.280	33.587
Observations	20,071	20,071	20,071	7,720	7,720
Unique households	6,268	6,268	6,268	$2,\!642$	$2,\!642$

Table 5: Accounting for Jefes y jefas – 2SLS Estimates

Note: Table 5 replicates Table 4, adding the female participation to the workfare program Jefes y jefas as a control. *** denotes significance at 1%, ** at 5%, and * at 10%. S.e. clustered by industry-occupation pairs are shown in parentheses. The sample used is the sample described in Section 5, that is, the sample of couples observed at least once in October 2001 or before, and once after the repeal of convertibility. In the last two columns, the sample is restricted to couples where women report working at least once before and after January 2002. Particip. is a female participation dummy, Empl. is a female employment dummy, Full T. is a binary variable equal to 1 if women report working 20 hours or more, Wish is a binary variable equal to 1 if women declare that they are seeking to work longer hours, Hours is a continuous variable measuring the weekly working hours. The time-varying controls are the household composition variables listed in the bottom panel of Table A1, as well as the amounts of household income listed in Table A2.

	Exte (A	nsive Margi .ll Women)	Intensive Margin (Working Women)		
	Δ Particip.	$\Delta \text{Empl.}$	Δ Full T.	ΔW ish	$\Delta Hours$
Partner enters unemployment	0.0685***	0.0414***	0.0158	0.164***	3.194***
	(0.017)	(0.016)	(0.014)	(0.031)	(0.959)
Partner exits unemployment	-0.00803	0.00168	-0.0121	0.0433	3.178***
	(0.020)	(0.019)	(0.016)	(0.036)	(1.152)
Region \times Time FE	Yes	Yes	Yes	Yes	Yes
Δ Time-varying controls	Yes	Yes	Yes	Yes	Yes
Observations	12,902	12,902	12,902	5,549	$5,\!549$
Unique households	5,834	$5,\!834$	$5,\!834$	3,019	3,019

Table 6: Asymmetric Effect of Male Employment on Participation- First Difference Estimates

Note: *** denotes significance at 1%, ** at 5%, and * at 10%. S.e. clustered by households are shown in parentheses. Δ indicates that outcomes and time-varying controls are measured in first difference. Entry is a binary variable equal to 1 if the male partner enters unemployment ($\Delta Unemployment^m = 1$), 0 otherwise, and Exit is a binary variable equal to 1 if he exits unemployment and finds a job ($\Delta Unemployment^m = -1$), 0 otherwise. The time-varying controls are the household composition variables listed in the bottom panel of Table A1, as well as the amounts of household income listed in Table A2. The sample used is the sample described in Section 5, that is, the sample of couples observed at least once in October 2001 or before, and once after the repeal of convertibility. Note that by construction, the number of observations is inferior to that in Table 4, due to the first differentiation.

		Simple Model			Mode	el with Interact	tions
	First Stage	2SLS: Part	2SLS: Emp	2SLS: Full	2SLS: Part	2SLS: Emp	2SLS: Full
Shock Intensity	-11.21^{***} (2.462)						
Male Partner's Log Lab. Inc.		-0.246***	-0.215**	0.0381			
		(0.082)	(0.086)	(0.049)			
Male Partner's Log Lab. Inc. \times Q1					-0.274***	-0.257***	0.0788
					(0.089)	(0.081)	(0.054)
Male Partner's Log Lab. Inc. \times Q2					-0.126*	-0.0514	0.0962
					(0.063)	(0.066)	(0.058)
Male Partner's Log Lab. Inc. \times Q3					0.0103	0.0178	0.124
					(0.052)	(0.067)	(0.120)
Male Partner's Log Lab. Inc. \times Q4					0.0208	0.0504	0.134**
					(0.052)	(0.048)	(0.065)
Male Partner's Log Lab. Inc. \times Q5					0.125^{**}	0.200***	0.288^{***}
					(0.058)	(0.060)	(0.100)
Household FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Region \times Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Industry \times Occupation FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time-varying controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Male Monthly Income before 2002	689.38						
Male Monthly Income in 2002	461.18						
$ar{Y}$		0.473	0.425	0.296	0.473	0.425	0.296
Observations	17,027	17,027	17,027	17,027	17,027	17,027	17,027
Unique households	5,545	5,545	5,545	5,545	5,545	5,545	$5,\!545$

Table 7: Female Participation and Male Income – First Stage & 2SLS Estimates

Note: The simple model measures the causal participation and employment responses to a negative income shocks affecting employed male partners. The first column displays the first stage estimates. The model with interaction disentangles the effect by quintile of the per capita household income distribution at baseline (first stage estimations not reported). *** denotes significance at 1%, ** at 5%, and * at 10%. S.e. clustered by industry-occupation pairs are shown in parentheses. The time-varying controls are the household composition variables listed in the bottom panel of Table A1, as well as the amounts of household income listed in Table A2. The sample used is the sample described in Section 5, that is, the sample of couples observed at least once in October 2001 or before, and once after the repeal of convertibility. Note that the number of observations is inferior to Table 4, since I documenting only men who are working over the period.

	Extensive Margin (All Women)					Intensive (Working	Margin Women)			
	Particip.	Particip.	Empl.	Empl.	Full T.	Full T.	Wish	Wish	Hours	Hours
Unemployed in t	0.0631***	0.0583***	0.0421***	* 0.0377**	* 0.0225***	* 0.0216***	* 0.0579***	* 0.0589**	** 0.0196	0.0878
	(0.007)	(0.007)	(0.007)	(0.007)	(0.006)	(0.006)	(0.014)	(0.014)	(0.396)	(0.388)
Unemployed in t-1	0.0181^{***}		0.0167^{**}		0.00334		-0.00418		-0.269	
	(0.007)		(0.007)		(0.006)		(0.013)		(0.383)	
Household FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Region \times Time FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Time-varying controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
\mathbb{R}^2	0.797	0.797	0.822	0.822	0.798	0.798	0.639	0.639	0.803	0.803
\overline{Y} if Husb. Employed	0.480	0.480	0.436	0.436	0.295	0.295	0.226	0.226	32.807	32.807
Observations	95,222	95,222	$95,\!222$	95,222	95,222	$95,\!222$	$36,\!136$	36,136	$36,\!136$	$36,\!136$
Unique households	39,278	39,278	39,278	39,278	39,278	39,278	$15,\!546$	$15,\!546$	$15,\!546$	$15,\!546$

Table 8: Delayed Female Labor Market Responses to Male Unemployment

Note: *** denotes significance at 1%, ** at 5%, and * at 10%. S.e. clustered by households are shown in parentheses. In the 'Extensive margin' columns, the full sample of couples is used. In the 'Intensive margin' column, the sample is restricted to couples where women report working at least once before and after January 2002. Note that by construction, the number of observations is inferior to Table 2, because of the lagged independent term. Particip. is a female participation dummy, Empl. is a female employment dummy, Full T. is a binary variable equal to 1 if women report working 20 hours or more, Wish is a binary variable equal to 1 if women declare that they are seeking to work longer hours, Hours is a continuous variable measuring the weekly working hours. Unemployed in t is equal to 1 if the male partner is unemployed, and Unemployed in t - 1 is equal to 1 if the male partner was unemployed during the previous period, 0 otherwise. The time-varying controls are the household composition variables listed in the bottom panel of Table A1, as well as the amounts of household income listed in Table A2.



Figure 1: Medium Run Labor Market Response: Coefficients \hat{a}^t

Note: Figure (a) shows estimates from a single regression, where the dependent variable is female participation, and each coefficient and corresponding 90% confidence interval captures the deviation in participation explained by the different intensity of the shock experienced by male partners after the repeal of the convertibility law, taking May 1996 as a reference (Equation 7). Figure (b) presents the same results with regard to female employment. The sample used is the main sample described in Section 3. In the last two columns, the sample is restricted to couples where the women are working. I estimate the model using pooled OLS with period-by-province fixed effects, and control for all the demographic characteristics and time-varying controls listed in Table Table A1, as well as for the amounts of household income listed in Table A2. Standard errors are two-way clustered by household, and by male industry-occupation pairs, because the shock exposure varies at the industry-occupation level.

A Appendix: Data & Baseline

	1996-1999	2000-2002	2003-2007
Female Demographics			
Age	39.02	39.23	39.49
	(9.75)	(9.83)	(9.90)
Level of education achieved			
No education	0.11	0.09	0.07
	(0.31)	(0.28)	(0.25)
Primary	0.50	0.48	0.42
	(0.50)	(0.50)	(0.49)
Secondary	0.27	0.28	0.32
	(0.44)	(0.45)	(0.47)
Tertiary	0.13	0.15	0.19
	(0.34)	(0.36)	(0.40)
Male Demographics			
Age	41.81	41.91	42.21
	(10.04)	(10.07)	(10.21)
Level of education achieved			
No education	0.11	0.09	0.07
	(0.31)	(0.29)	(0.26)
Primary	0.53	0.52	0.47
	(0.50)	(0.50)	(0.50)
Secondary	0.25	0.27	0.31
	(0.43)	(0.45)	(0.46)
Tertiary	0.11	0.12	0.15
	(0.31)	(0.32)	(0.36)
Household Demographics			
Number of children 0-2	0.18	0.17	0.15
	(0.40)	(0.39)	(0.37)
Number of children 3-5	0.29	0.27	0.25
	(0.53)	(0.52)	(0.49)
Number of children 6-15	0.97	0.99	0.88
	(1.13)	(1.17)	(1.08)
Number of children 15+ studying	0.36	0.40	0.38
	(0.68)	(0.72)	(0.71)
Number of adults 16-64	0.42	0.41	0.40
	(0.83)	(0.83)	(0.81)
Number of adults 65+	0.05	0.05	0.04
	(0.24)	(0.23)	(0.20)
Observations	89,475	53,751	100,014

Table A1: Summary Statistics: Demographics

Note: The sample used is the main sample described in Section 3, divided into three periods. The core analysis focuses on the period 2000-2002.

	1996-1999	2000-2002	2003-2007
Household income			
Other Labor Income > 0	0.23	0.21	0.21
	(0.42)	(0.41)	(0.41)
Pensions > 0	0.06	0.06	0.06
	(0.25)	(0.23)	(0.23)
Capital > 0	0.01	0.01	0.02
	(0.11)	(0.11)	(0.15)
Unemployment > 0	0.01	0.01	0.01
	(0.10)	(0.12)	(0.10)
Remittances > 0	0.01	0.02	0.02
	(0.12)	(0.15)	(0.15)
Other > 0	0.01	0.03	0.08
	(0.12)	(0.18)	(0.27)
Observations	89,475	53,751	100,014
Amount when positive			
Other Labor Income	564.73	457.95	447.15
	(525.21)	(434.53)	(393.92)
Pensions	318.83	299.86	294.59
	(314.29)	(286.67)	(315.22)
Capital	580.36	510.62	495.64
	(657.85)	(906.54)	(2, 325.10)
Unemployment	769.26	373.57	508.26
	(2, 817.77)	(740.02)	(1,534.14)
Remittances	197.90	160.95	169.59
	(191.84)	(177.88)	(401.39)
Other	340.43	218.10	142.81
	(1, 349.30)	(792.86)	(746.06)

Table A2: Summary Statistics: Income Sources

Note: The sample used is the main sample described in Section 3, divided into three periods. The core analysis focuses on the period 2000-2002.

	All wor	nen	Working one	ce or more
	Not working	Working	Discontinuously	Continuously
Own Demographics				
Age	38.318	39.128***	37.972	39.454***
Level of education achieved				
No education	0.118	0.064^{***}	0.109	0.049***
Primary	0.547	0.352***	0.525	0.299^{***}
Secondary	0.280	0.309***	0.280	0.319***
Tertiary	0.054	0.275***	0.087	0.332***
Partner's Demographics				
Age	41.358	41.735***	40.901	41.968***
Level of education achieved				
No education	0.115	0.077***	0.113	0.065***
Primary	0.556	0.429***	0.556	0.389***
Secondary	0.258	0.327***	0.256	0.351***
Tertiary	0.072	0.166***	0.075	0.195***
Household composition				
N children 0-2	0.209	0.141***	0.175	0.131***
N children 3-5	0.341	0.252***	0.308	0.238***
N children 6-15	1.097	0.987^{***}	1.131	0.940***
N children 15+ studying	0.372	0.398***	0.369	0.407***
N adults 16-64	0.455	0.348***	0.438	0.317***
N adults 65+	0.040	0.041	0.039	0.042**
Household income				
Other Labor Income > 0	0.206	0.189^{***}	0.211	0.176^{***}
Pensions > 0	0.074	0.051^{***}	0.059	0.050***
Capital > 0	0.015	0.019***	0.014	0.020***
Unemployment > 0	0.010	0.007***	0.012	0.006***
Remittances > 0	0.019	0.014^{***}	0.021	0.012***
Other > 0	0.057	0.040***	0.069	0.034***
Observations	131,138	112,102	58,385	83,815

Table A3: Summary Statistics (Means) on Female Partners, by Employment Status

Note: *** denotes significance at 1% of the difference in means between women working/not working (col 1-2), and then, using the panel dimension, between women continuously employed and women temporarily employed (col 3-4). ** denotes the difference at 5%, and * at 10%. The sample used is the main sample described in Section 3.

	1996-1999	2000-2002	2003-2007
Female Demographics			
Age	37.98	38.53	39.41
	(9.48)	(9.69)	(9.97)
Level of education achieved			
No education	0.12	0.10	0.07
	(0.32)	(0.30)	(0.25)
Primary	0.49	0.47	0.41
	(0.50)	(0.50)	(0.49)
Secondary	0.26	0.28	0.33
	(0.44)	(0.45)	(0.47)
Tertiary	0.12	0.14	0.19
	(0.33)	(0.35)	(0.39)
Male Demographics			
Age	40.87	41.34	42.23
	(9.70)	(9.90)	(10.26)
Level of education achieved			
No education	0.12	0.10	0.08
	(0.32)	(0.31)	(0.27)
Primary	0.53	0.51	0.46
	(0.50)	(0.50)	(0.50)
Secondary	0.26	0.28	0.32
	(0.44)	(0.45)	(0.47)
Tertiary	0.10	0.10	0.14
	(0.29)	(0.30)	(0.35)
Household Demographics			
Number of children 0-2	0.20	0.18	0.15
	(0.43)	(0.41)	(0.38)
Number of children 3-5	0.34	0.30	0.26
	(0.58)	(0.54)	(0.50)
Number of children 6-15	1.12	1.10	0.95
	(1.21)	(1.22)	(1.14)
Number of children 15+ studying	0.35	0.39	0.41
	(0.68)	(0.72)	(0.74)
Number of adults 16-64	0.40	0.41	0.42
	(0.82)	(0.84)	(0.85)
Number of adults $65+$	0.04	0.04	0.04
	(0.22)	(0.21)	(0.20)
Observations	89,475	53,751	100,014

Table A4: Summary Statistics: Demographics (Unweighted)

Note: The sample used is the main sample described in Section 3, divided into three periods. The core analysis focuses on the period 2000-2002.

	Extensive Margin (All Women)			Intensive Margin (Working Women)	
	Particip.	Empl.	Full T.	Wish	Hours
Male partner is unemployed	0.0536***	0.0293***	0.0142***	0.0518***	0.0782
	(0.004)	(0.004)	(0.004)	(0.008)	(0.238)
Household FE	Yes	Yes	Yes	Yes	Yes
Region \times Time FE	Yes	Yes	Yes	Yes	Yes
\mathbb{R}^2	0.762	0.791	0.755	0.585	0.755
\bar{Y} if Husb. Employed	0.503	0.457	0.308	0.209	32.383
Observations	$243,\!240$	$243,\!240$	$243,\!240$	$101,\!332$	$101,\!332$
Unique households	81,956	$81,\!956$	$81,\!956$	$36,\!614$	$36,\!614$

 Table A5: Baseline Specification (Without Time-varying Co-variates)

Note: *** denotes significance at 1%, ** at 5%, and * at 10%. S.e. clustered by households are shown in parentheses. Particip. is a female participation dummy, Empl. is a female employment dummy, Full T. is a binary variable equal to 1 if women report working 20 hours or more, Wish is a binary variable equal to 1 if women declare that they are seeking to work longer hours, Hours is a continuous variable measuring the weekly working hours. The sample used is the main sample described in Section 3. In the last two columns, the sample is restricted to couples where the women are working.

B Appendix: Instrument

Industries in the EPH – Over 1996–2007, *INDEC* used two different classifications to characterize industries in the *EPH*.

- Between 1996 and 2002, the survey relied on the 3-digit ISIC Rev.3 classification provided by the ILO.
- Between 2003 and 2007, the survey turned to a 4-digit structure and adopted the *CAES* Mercosur 2000 classification.

In this paper, I make use of the first 2 digits of both industry classifications. The main advantage is that they correspond to exactly the same broad industries across classifications, except those for the wholesale and retail trade, which are merged in the *CAES* Mercosur 2000 (INDEC, 2011). Therefore, in the core of the paper, I consider the following 22 industries: (1) Agriculture, (2) Food industry, (3) Textile, (4) Chemistry, (5) Metal, (6) Other goods, (7) Energy/water, (8) Construction, (9) Wholesale trade, (10) Retail trade, (11) Accommodation/food services, (12) Transportation, (13) Communication, (14) Finance –Insurance, (15) Real estate, (16) Public administration, (17) Education, (18) Health, (19) Other social-local services, (20) Repair, (21) Domestic services, (22) Other services. When I turn to the dynamics of participation, I merge sectors (9) and (10) into a single 'Trade' sector.

Occupations in the EPH – Over 1996-2007, *INDEC* used three different classifications to define occupations, which differed in many ways from the international standard ISCO. All these classifications have in common that they are multidimensional, and notably characterize the qualifications for a given job. Qualification reflects the complexity of the tasks in each job. It is measured according to the combination of activities or actions performed, the instruments used, and the management of raw materials. It reflects the knowledge and skills required to perform the tasks, and therefore helps to assess the degree of complexity of an occupation. Since its origin, qualification in the *EPH* has been coded as a number between 1 and 4, and its significance has remained remarkably stable over time.

• Originally, the *Clasificador de Ocupaciones* of the *EPH* (*CO-EPH*) was *INDEC*'s own classificatory instrument for occupations. This two-digit classification was bidimensional and assessed the purpose of the work (character) and the complexity of the tasks performed (qualification). Qualification had four categories: professional, qualified, semi-qualified and unqualified (INDEC, 2000).

- Next, the *Clasificador Nacional de Ocupaciones 1991* (*CNO-91*) was introduced with a double objective: to update and unify the measurement of occupations and to ease international comparisons, notably with ISCO. The new classification combined four dimensions (general character, technology, hierarchy and complexity) in a three-digit structure. The last digit measured qualification in four categories: professional, technical, clerical/operational and elementary. The classification was gradually implemented until October 1997. In May 1996, the most populated provinces had already adopted the *CNO-91* instead of *CO-EPH*.³¹ The remaining provinces opted for *CNO-91* between 1996 and 1997 (INDEC, 2000).³²
- In 2001, the *Clasificador Nacional de Ocupaciones 2001 (CNO-01)* further disentangled the types of occupations with a 5-digit structure (INDEC, 2001). The last digit, qualification, kept the same definition as in *CNO-91*. All the provinces adopted the CNO-01 in 2003.

In this paper, following *INDEC*'s recommendations (INDEC, 2000), I adopt the following simple equivalence:

CO- EPH	<i>CNO-91</i>	<i>CNO-01</i>	Digit
Professional	Professional	Professional	1
Qualified	Technical	Technical	2
Semi-qualified	Clerical/operational	Clerical/operational	3
Unqualified	Elementary	Elementary	4



Figure B1: Trend Coefficients $\hat{a}^{i,o}$

Note: The sample used consists of men of working age in urban Argentina between 1996 and 2002. For each industry-occupation pair, I run the following regression: $E_{j,r,t}^{i,o} = a^{i,o}Trend_{j,t} + b^{i,o}Post_t + \phi_r^{i,o} + \epsilon_{j,r,t}^{i,o}$. The figure displays $\hat{a}^{i,o}$ estimates from these separate regressions, and their corresponding 90% confidence interval. The significantly negative coefficients are displayed in black, the significantly positive coefficients in light grey, and the insignificant coefficients in grey.



Figure B2: Shock Coefficients $\hat{b}^{i,o}$

Note: The sample used consists of men of working age in urban Argentina between 1996 and 2002. For each industry-occupation pair, I run the following regression: $E_{j,r,t}^{i,o} = a^{i,o}Trend_{j,t} + b^{i,o}Post_t + \phi_r^{i,o} + \epsilon_{j,r,t}^{i,o}$. The figure displays $\hat{b}^{i,o}$ estimates from these separate regressions, and their corresponding 90% confidence interval. The significantly negative coefficients are displayed in black, the significantly positive coefficients in light grey, and the insignificant coefficients in grey.



Figure B3: Partners' Industry in October 2001

Note: Figure B3 displays the frequency of each possible combination of industry-occupation pairs for couples where both partners are active. Industries: (1) Agriculture, (2) Food industry, (3) Textile, (4) Chemistry, (5) Metal, (6) Other goods, (7) Energy/water, (8) Construction, (9) Wholesale trade, (10) Retail trade, (11) Accommodation/food services, (12) Transportation, (13) Communication, (14) Finance – Insurance, (15) Real estate, (16) Public administration, (17) Education, (18) Health, (19) Other social-local services, (20) Repair, (21) Domestic services, (22) Other services.



Figure B4: Partners' Future Shock given their Industry in October 2001

Note: Figure B4 plots the possible combinations of male-female employment shocks for couples where both partners are active, together with their share in my sample. The plain line displays the slope of a regression of women's shock intensity on their partner's shock intensity, weighted by their share. The dashed line displays the slope obtained from the same regression, but excluding the most prevalent combination – construction operators living with unqualified domestic workers.

	Extensive Margin (All Women)	Intensive Margin (Working Women)		
	Male partner is unemployed	Male partner is unemployed		
Shock Intensity	5.611***	4.441***		
	(0.486)	(0.815)		
Composition				
N children 0-2	0.00117	0.0338^{*}		
	(0.012)	(0.020)		
N children 3-5	-0.0141	0.00374		
	(0.016)	(0.036)		
N children 6-15	-0.0109	0.00737		
	(0.011)	(0.024)		
N children 15+ studying	-0.00777	-0.00916		
	(0.014)	(0.013)		
N adult 16-64	-0.00182	-0.00677		
	(0.011)	(0.015)		
N adult 65+	0.0765***	0.0165		
	(0.029)	(0.063)		
Household Income				
Labor	-0.00426**	-0.00679**		
	(0.002)	(0.003)		
Pensions	0.00109	0.000818		
	(0.005)	(0.007)		
Capital	0.00835**	0.00920		
	(0.004)	(0.008)		
Unemployment	0.0697***	0.0742***		
	(0.007)	(0.018)		
Remittances	0.0446***	0.0113		
	(0.006)	(0.010)		
Other	0.0307***	0.0193***		
	(0.005)	(0.005)		
Household FE	Yes	Yes		
Region \times Time FE	Yes	Yes		
Industry \times Occupation FE	Yes	Yes		
KP	133.13	29.68		
Mean(Intensity)	0.00153	0.00130		
SD(Intensity)	0.00499	0.00464		
Observations	20,071	7,720		
Unique households	6,268	$2,\!642$		

Table B1: First Stage Estimates

Note: See Table 3.

	Extensive Margin (All Women)		Intensive (Working V	Margin Vomen)	
	Particip.	Empl.	Full T.	Wish	Hours
Male partner is unemployed	0.484***	0.492***	-0.0677	0.543**	-12.63
	(0.085)	(0.097)	(0.076)	(0.268)	(9.132)
Composition					
N children 0-2	0.0109	0.00996	0.0203^{*}	0.0209	3.165**
	(0.014)	(0.012)	(0.012)	(0.027)	(1.059)
N children 3-5	0.0238	0.0249^{*}	0.0246^{*}	0.0248	2.286^{*}
	(0.015)	(0.013)	(0.014)	(0.037)	(1.243)
N children 6-15	0.0252	0.0267^{*}	0.0109	0.0172	0.137
	(0.016)	(0.014)	(0.010)	(0.027)	(1.071)
N children 15+ studying	0.0146	0.0252	0.00984	0.0373	0.420
	(0.018)	(0.016)	(0.011)	(0.023)	(0.923)
N adult 16-64	0.00761	0.0124	-0.00114	0.0564^{***}	-0.484
	(0.012)	(0.010)	(0.008)	(0.019)	(0.835)
N adult 65+	0.0192	-0.0133	-0.0135	-0.0282	0.837
	(0.051)	(0.045)	(0.031)	(0.089)	(2.273)
Household Income					
Labor	0.0116***	0.0102***	0.00240	-0.00185	-0.104
	(0.002)	(0.003)	(0.002)	(0.004)	(0.187)
Pensions	-0.0181***	* -0.0174***	* -0.0142***	-0.00775	-0.482
	(0.005)	(0.006)	(0.004)	(0.012)	(0.384)
Capital	0.00686	0.00811^{*}	0.00393	-0.0200**	-0.0181
	(0.006)	(0.005)	(0.004)	(0.010)	(0.351)
Unemployment	-0.0377***	* -0.0461***	* 0.00101	-0.0611***	1.298*
	(0.007)	(0.007)	(0.008)	(0.023)	(0.669)
Remittances	-0.0230***	* -0.0360***	* -0.0126**	0.0143	-0.214
	(0.007)	(0.008)	(0.006)	(0.018)	(0.505)
Other	-0.0235***	* -0.0273***	*-0.00434	-0.00830	-0.118
	(0.006)	(0.008)	(0.004)	(0.009)	(0.293)
Household FE	Yes	Yes	Yes	Yes	Yes
Region \times Time FE	Yes	Yes	Yes	Yes	Yes
Industry \times Occupation FE	Yes	Yes	Yes	Yes	Yes
\overline{Y} if Husb. Employed	0.475	0.426	0.296	0.280	33.587
Observations	20,071	20,071	20,071	7,720	7,720
Unique households	6,268	6,268	6,268	2,642	2,642

Table B2: 2SLS Estimates

Note: See Table 4.

C Appendix: Robustness & Discussion



(a) Second Stage Results – Participation Re-(b) First Stage Results – Job Loss Response to sponse to Male Job LossShock

Figure C1: 2SLS Results, Excluding one Industry at a Time

Note: The figures show estimates from 23 different regressions. Each coefficient and its corresponding 90% confidence interval come from a separate regression, where I estimate the set of equations (3) excluding each of the 22 industries one by one. For each estimation, the excluded industry is displayed on the y-axis. As a benchmark, I also report the main estimates of Tables 3 and 4, where none of the industries is excluded. The sample used is the sample described in Section 5, that is, the sample of couples observed at least once in October 2001 or before, and once after the repeal of convertibility.



Figure C2: Response of Male Unemployment to the 2002 Shock Intensity: $\hat{\pi}^y$

Note: Figure C2 shows estimates from a single regression, where the dependent variable is male unemployment, and each coefficient and corresponding 90% confidence interval captures the deviation in unemployment explained by the different intensity of the shock experienced by male partners after the repeal of the convertibility law, taking May 1996 as a reference (Equation 4). Circles (diamonds) denote differential time effects before (after) January 2002. The specification controls for household fixed effects, period-by-province fixed effects, and industry-occupation fixed effects. S.e. clustered by industry-occupation pairs are shown in parentheses. The time-varying controls are the household composition variables listed in the bottom panel of Table A1, as well as the amounts of household income listed in Table A2. The sample used is the sample described in Section 5, that is, the sample of couples observed at least once in October 2001 or before, and once after the repeal of convertibility.

	Extensive Margin (All Women)		Intensive (Working	Margin Women)	
	Particip.	Empl.	Full T.	Hours	Wish
Male partner is unemployed	0.0607***	0.0457***	0.0219**	0.659	0.0699***
	(0.012)	(0.011)	(0.010)	(0.611)	(0.023)
Composition					
N children 0-2	0.0133	0.0119	0.0208**	2.709***	0.0282
	(0.013)	(0.012)	(0.010)	(0.859)	(0.027)
N children 3-5	0.0165	0.0179	0.0256^{**}	2.344**	0.0234
	(0.015)	(0.013)	(0.012)	(0.956)	(0.031)
N children 6-15	0.0196	0.0209*	0.0120	0.250	0.0178
	(0.013)	(0.012)	(0.011)	(0.863)	(0.026)
N children 15+ studying	0.0109	0.0215^{*}	0.0103	0.685	0.0322
	(0.012)	(0.011)	(0.011)	(0.790)	(0.024)
N adult 16-64	0.00777	0.0122	-0.00189	-0.288	0.0547***
	(0.010)	(0.010)	(0.009)	(0.745)	(0.020)
N adult 65+	0.0507	0.0179	-0.0190	0.400	-0.0140
	(0.039)	(0.037)	(0.030)	(1.965)	(0.080)
Household Income					
Labor	0.00952***	* 0.00826***	*0.00296	-0.0207	-0.00532
	(0.002)	(0.002)	(0.002)	(0.150)	(0.004)
Pensions	-0.0187***	· -0.0174***	* -0.0148***	* -0.455	-0.00796
	(0.005)	(0.005)	(0.004)	(0.345)	(0.011)
Capital	0.0101*	0.0118**	0.00340	-0.147	-0.0168*
	(0.005)	(0.005)	(0.004)	(0.338)	(0.009)
Unemployment	-0.00786	-0.0144***	*-0.00529	0.330	-0.0273**
	(0.005)	(0.005)	(0.005)	(0.318)	(0.010)
Remittances	-0.00457	-0.0163***	* -0.0164***	* -0.338	0.0169
	(0.006)	(0.005)	(0.004)	(0.426)	(0.012)
Other	-0.0102***	· -0.0133***	*-0.00696**	*-0.313*	0.000238
	(0.003)	(0.003)	(0.002)	(0.172)	(0.007)
Household FE	Yes	Yes	Yes	Yes	Yes
Region \times Time FE	Yes	Yes	Yes	Yes	Yes
\mathbb{R}^2	0.731	0.764	0.760	0.767	0.597
\bar{Y} if Husb. Employed	0.475	0.426	0.296	33.587	0.280
Observations	20,071	20,071	20,071	7,720	7,720
Unique households	6,268	6,268	6,268	2,642	2,642

Table C1: Baseline Specification – Replication
--

Note: This table replicates Table 2 with the subsample of couples observed around the repeal of convertibility, used in the core results section (Section 5). *** denotes significance at 1%, ** at 5%, and * at 10%. S.e. clustered by households are shown in parentheses.

	Extensive Margin (All Women)			Intensive Margin (Working Women)	
	Particip.	Empl.	Full T.	Wish	Hours
Male partner is unemployed	0.506^{***} (0.095)	0.524^{***} (0.103)	(0.0474)	0.629^{**} (0.307)	-12.89 (10.574)
Household FE	Yes	Yes	Yes	Yes	Yes
Region \times Time FE	Yes	Yes	Yes	Yes	Yes
Industry \times Occupation FE	Yes	Yes	Yes	Yes	Yes
\bar{Y} if Husb. Employed	0.437	0.384	0.255	0.303	32.291
Observations	18,757	18,757	18,757	$6,\!466$	6,466
Unique households	$5,\!870$	$5,\!870$	$5,\!870$	2,244	2,244

Table C2: 2SLS Estimates – Excluding Same Industry Couples

Note: *** denotes significance at 1%, ** at 5%, and * at 10%. S.e. clustered by industry-occupation pairs are shown in parentheses. Particip. is a female participation dummy, Empl. is a female employment dummy, Full T. is a binary variable equal to 1 if women report working 20 hours or more, Wish is a binary variable equal to 1 if women declare that they are seeking to work longer hours, Hours is a continuous variable measuring the weekly working hours. The sample used is the sample described in Section 5, that is, the sample of couples observed at least once in October 2001 or before, and once after the repeal of convertibility. In the last two columns, the sample is restricted to couples where women report working at least once before and after January 2002. In addition, for all estimations, I exclude couples working (or being unemployed after a job in the same industry) at least once before and after convertibility ends. The time-varying controls are the household composition variables listed in the bottom panel of Table A1, as well as the amounts of household income listed in Table A2.

	Dep. var. = 1 if household drops out in $t+1$			
	(1)	(2)	(3)	(4)
2002 Shock intensity	0.119	-0.171	0.363	-0.208
	(0.454)	(0.250)	(0.484)	(0.301)
Composition				
N children 0-2	0.0151^{*}	0.00511	0.0280***	-0.0127***
	(0.007)	(0.004)	(0.008)	(0.005)
N children 3-5	0.0226^{**}	-0.00224	0.0463***	-0.0183***
	(0.010)	(0.003)	(0.011)	(0.004)
N children 6-15	0.0285***	-0.00484***	0.0381***	-0.0200***
	(0.008)	(0.001)	(0.008)	(0.002)
N children 15+ studying	0.0119^{*}	-0.000650	0.00670	-0.0119***
	(0.006)	(0.002)	(0.006)	(0.004)
N adult 16-64	0.00486	-0.00323*	0.00214	-0.00502
	(0.005)	(0.002)	(0.007)	(0.003)
N adult 65+	-0.0239	-0.0181**	-0.0407	-0.0380***
	(0.022)	(0.008)	(0.024)	(0.008)
Household Income		. ,		. ,
Labor	0.00219**	0.000649	0.000877	-0.00157
	(0.001)	(0.001)	(0.001)	(0.001)
Pensions	-0.00127	-0.000153	-0.00267	0.000134
	(0.003)	(0.001)	(0.003)	(0.001)
Capital	-0.00101	-0.000358	0.00296	0.00344
-	(0.003)	(0.002)	(0.004)	(0.004)
Unemployment	-0.00123	-0.00265	-0.00309	0.0000378
	(0.004)	(0.004)	(0.002)	(0.003)
Remittances	0.00177	0.00210	-0.000147	0.00618**
	(0.003)	(0.003)	(0.003)	(0.003)
Other	-0.000864	-0.000182	0.000239	0.00185
	(0.001)	(0.001)	(0.002)	(0.001)
Region \times Time FE	Yes	Yes	Yes	Yes
Household FE	Yes	No	Yes	No
Demog. Controls	No	Yes	No	Yes
\bar{Y}	0.062	0.062	0.085	0.179
Proportion of couples interviewed < 4	0.200	0.200	0.260	0.369
Observations	20,071	20,071	35,771	47,654

Table C3: Panel Attrition

Note: Columns (1) and (2) of Table C3 display estimates based on the main sample described in Section 5. Column (1) presents the results of an estimation where the probability of dropping out of the sample depends on within-household variation in the exposure to the shock intensity measured in 2002, controlling for time-varying household characteristics, as well as household and province-by-period fixed effects. Column (2) presents the results of a pooled OLS regression, without the household fixed effects, but with standard demographic controls: the age of both spouses in years and their education split into 4 stages: none, primary, secondary, and tertiary. The probability of dropping out depends on the variation in the exposure to the shock intensity measured in 2002 across households. Columns (3) and (4) take the attrition issue upstream, and consider the sample of households surveyed between October 2000 and October 2002, before I imposed the condition that partners should be observed at least once in October 2001 or before, and once after. The difference in the number of observations comes from the fact that singleton households are excluded from the regression in Column (3), but not from that in Column (4).

	Extensive Margin (All Women)	Intensive Margin (Working Women)		
	Male partner is unemployed	Male partner is unemployed		
Shock Intensity	5.578***	4.563***		
	(0.489)	(0.891)		
Workfare Beneficiary	0.00902	-0.0286		
	(0.019)	(0.038)		
Household FE	Yes	Yes		
Region \times Time FE	Yes	Yes		
Industry \times Occupation FE	Yes	Yes		
Time-varying controls	Yes	Yes		
F-stat	130.02	26.24		
Mean(Intensity)	0.00153	0.00130		
SD(Intensity)	0.00499	0.00464		
Observations	20,071	7,720		
Unique households	6,268	2,642		

Table C4: Accounting for Jefes y jefas – First Stage Estimates

Note: Table C4 replicates Table 3, adding the female participation to the workfare program Jefes y jefas as a control. *** denotes significance at 1%, ** at 5%, and * at 10%. S.e. clustered by industry-occupation pairs are shown in parentheses. The sample used is the sample described in Section 5, that is, the sample of couples observed at least once in October 2001 or before, and once after the repeal of convertibility. In the last two columns, the sample is restricted to couples where women report working at least once before and after January 2002. The time-varying controls are the household composition variables listed in the bottom panel of Table A1, as well as the amounts of household income listed in Table A2.